\begin{thebibliography}{78}
\expandafter\ifx\csname natexlab\endcsname\relax\def\natexlab#1{#1}\fi

\bibitem[{{Archontis} {et~al.}(2004){Archontis}, {Moreno-Insertis},
  {Galsgaard}, {Hood}, \& {O'Shea}}]{Arc04}
{Archontis}, V., {Moreno-Insertis}, F., {Galsgaard}, K., {Hood}, A., \&
  {O'Shea}, E. 2004, \aap, 426, 1047

\bibitem[{{Archontis} {et~al.}(2005){Archontis}, {Moreno-Insertis},
  {Galsgaard}, \& {Hood}}]{Arc05}
{Archontis}, V., {Moreno-Insertis}, F., {Galsgaard}, K., \& {Hood}, A.~W. 2005,
  \apj, 635, 1299

\bibitem[{{Bhat} \& {Brandenburg}(2016)}]{BhatBrandenburg15}
{Bhat}, P. \& {Brandenburg}, A. 2016, \aap, 587, A90

\bibitem[{{Birch} {et~al.}(2016){Birch}, {Schunker}, {Braun}, {Cameron},
  {Gizon}, {Lo{\"o}ptien}, \& {Rempel}}]{BSBCGLR16}
{Birch}, A.~C., {Schunker}, H., {Braun}, D.~C., {et~al.} 2016, Sci. Adv., 2,
  e1600557

\bibitem[{{Bourdin} \& {Brandenburg}(2018)}]{BB18}
{Bourdin}, P.-A. \& {Brandenburg}, A. 2018, ApJL, submitted, arXiv:1804.04160

\bibitem[{{Brandenburg}(2005)}]{Bra05}
{Brandenburg}, A. 2005, \apj, 625, 539

\bibitem[{{Brandenburg}(2016)}]{Bra15}
{Brandenburg}, A. 2016, \apj, 832, 6

\bibitem[{{Brandenburg} \& {Campbell}(1997)}]{BC97}
{Brandenburg}, A. \& {Campbell}, C. 1997, in Lecture Notes in Physics, Berlin
  Springer Verlag, Vol. 487, Accretion Disks - New Aspects, ed.
  E.~{Meyer-Hofmeister} \& H.~{Spruit}, 109

\bibitem[{{Brandenburg} {et~al.}(2014){Brandenburg}, {Gressel}, {Jabbari},
  {Kleeorin}, \& {Rogachevskii}}]{BGJKR14}
{Brandenburg}, A., {Gressel}, O., {Jabbari}, S., {Kleeorin}, N., \&
  {Rogachevskii}, I. 2014, \aap, 562, A53

\bibitem[{{Brandenburg} {et~al.}(2011){Brandenburg}, {Kemel}, {Kleeorin},
  {Mitra}, \& {Rogachevskii}}]{BKKMR11}
{Brandenburg}, A., {Kemel}, K., {Kleeorin}, N., {Mitra}, D., \& {Rogachevskii},
  I. 2011, \apjl, 740, L50

\bibitem[{{Brandenburg} {et~al.}(2012){Brandenburg}, {Kemel}, {Kleeorin}, \&
  {Rogachevskii}}]{BKKR12}
{Brandenburg}, A., {Kemel}, K., {Kleeorin}, N., \& {Rogachevskii}, I. 2012,
  \apj, 749, 179

\bibitem[{{Brandenburg} {et~al.}(2013){Brandenburg}, {Kleeorin}, \&
  {Rogachevskii}}]{BKR2013}
{Brandenburg}, A., {Kleeorin}, N., \& {Rogachevskii}, I. 2013, \apjl, 776, L23

\bibitem[{{Brandenburg} {et~al.}(1995){Brandenburg}, {Nordlund}, {Stein}, \&
  {Torkelsson}}]{BNST95}
{Brandenburg}, A., {Nordlund}, A., {Stein}, R.~F., \& {Torkelsson}, U. 1995,
  \apj, 446, 741

\bibitem[{{Brandenburg} {et~al.}(2016){Brandenburg}, {Rogachevskii}, \&
  {Kleeorin}}]{BRK2016}
{Brandenburg}, A., {Rogachevskii}, I., \& {Kleeorin}, N. 2016, New J. Phys.,
  18, 125011

\bibitem[{{Chen} {et~al.}(2017){Chen}, {Rempel}, \& {Fan}}]{CRF17}
{Chen}, F., {Rempel}, M., \& {Fan}, Y. 2017, \apj, 846, 149

\bibitem[{{Cheung} \& {Isobe}(2014)}]{CI14}
{Cheung}, M.~C.~M. \& {Isobe}, H. 2014, Liv. Rev. Solar Phys., 11, 3

\bibitem[{{Cheung} {et~al.}(2010){Cheung}, {Rempel}, {Title}, \&
  {Sch{\"u}ssler}}]{CRTS10}
{Cheung}, M.~C.~M., {Rempel}, M., {Title}, A.~M., \& {Sch{\"u}ssler}, M. 2010,
  \apj, 720, 233

\bibitem[{{Choudhuri} \& {Gilman}(1987)}]{CG87}
{Choudhuri}, A.~R. \& {Gilman}, P.~A. 1987, \apj, 316, 788

\bibitem[{{Evershed}(1909)}]{Eve09}
{Evershed}, J. 1909, \mnras, 69, 454

\bibitem[{{Fan}(2001)}]{Fan01}
{Fan}, Y. 2001, \apjl, 554, L111

\bibitem[{{Fan}(2009)}]{Fan09}
{Fan}, Y. 2009, Living Reviews in Solar Physics, 6, 4

\bibitem[{{Fan} \& {Fang}(2014)}]{Fan2014}
{Fan}, Y. \& {Fang}, F. 2014, \apj, 789, 35

\bibitem[{{Fan} {et~al.}(1993){Fan}, {Fisher}, \& {Deluca}}]{FFD93}
{Fan}, Y., {Fisher}, G.~H., \& {Deluca}, E.~E. 1993, \apj, 405, 390

\bibitem[{{Fournier} {et~al.}(2017){Fournier}, {Arlt}, {Ziegler}, \&
  {Strassmeier}}]{FAZS17}
{Fournier}, Y., {Arlt}, R., {Ziegler}, U., \& {Strassmeier}, K.~G. 2017, \aap,
  607, A1

\bibitem[{{Galloway} \& {Weiss}(1981)}]{GW81}
{Galloway}, D.~J. \& {Weiss}, N.~O. 1981, \apj, 243, 945

\bibitem[{{Greer} {et~al.}(2015){Greer}, {Hindman}, {Featherstone}, \&
  {Toomre}}]{Greer15}
{Greer}, B.~J., {Hindman}, B.~W., {Featherstone}, N.~A., \& {Toomre}, J. 2015,
  \apjl, 803, L17

\bibitem[{{Hanasoge} {et~al.}(2012){Hanasoge}, {Duvall}, \&
  {Sreenivasan}}]{HDS12}
{Hanasoge}, S.~M., {Duvall}, T.~L., \& {Sreenivasan}, K.~R. 2012, Proc. Natl
  Acad. Sci., 109, 11928

\bibitem[{{Jabbari} {et~al.}(2015){Jabbari}, {Brandenburg}, {Kleeorin},
  {Mitra}, \& {Rogachevskii}}]{Jabbari15}
{Jabbari}, S., {Brandenburg}, A., {Kleeorin}, N., {Mitra}, D., \&
  {Rogachevskii}, I. 2015, \apj, 805, 166

\bibitem[{{Jabbari} {et~al.}(2017){Jabbari}, {Brandenburg}, {Kleeorin}, \&
  {Rogachevskii}}]{Jabbari2017}
{Jabbari}, S., {Brandenburg}, A., {Kleeorin}, N., \& {Rogachevskii}, I. 2017,
  \mnras, 467, 2753

\bibitem[{{Jabbari} {et~al.}(2014){Jabbari}, {Brandenburg}, {Losada},
  {Kleeorin}, \& {Rogachevskii}}]{Jabbari2014}
{Jabbari}, S., {Brandenburg}, A., {Losada}, I.~R., {Kleeorin}, N., \&
  {Rogachevskii}, I. 2014, \aap, 568, A112

\bibitem[{{Jabbari} {et~al.}(2016){Jabbari}, {Brandenburg}, {Mitra},
  {Kleeorin}, \& {Rogachevskii}}]{Jabbari16}
{Jabbari}, S., {Brandenburg}, A., {Mitra}, D., {Kleeorin}, N., \&
  {Rogachevskii}, I. 2016, \mnras, 459, 4046

\bibitem[{{K{\"a}pyl{\"a}} {et~al.}(2016){K{\"a}pyl{\"a}}, {Brandenburg},
  {Kleeorin}, {K{\"a}pyl{\"a}}, \& {Rogachevskii}}]{KBKKR16}
{K{\"a}pyl{\"a}}, P.~J., {Brandenburg}, A., {Kleeorin}, N., {K{\"a}pyl{\"a}},
  M.~J., \& {Rogachevskii}, I. 2016, \aap, 588, A150

\bibitem[{{K{\"a}pyl{\"a}} {et~al.}(2012){K{\"a}pyl{\"a}}, {Brandenburg},
  {Kleeorin}, {Mantere}, \& {Rogachevskii}}]{KaBKMR12}
{K{\"a}pyl{\"a}}, P.~J., {Brandenburg}, A., {Kleeorin}, N., {Mantere}, M.~J.,
  \& {Rogachevskii}, I. 2012, \mnras, 422, 2465

\bibitem[{{K{\"a}pyl{\"a}} {et~al.}(2013){K{\"a}pyl{\"a}}, {Brandenburg},
  {Kleeorin}, {Mantere}, \& {Rogachevskii}}]{KapyBeijing12}
{K{\"a}pyl{\"a}}, P.~J., {Brandenburg}, A., {Kleeorin}, N., {Mantere}, M.~J.,
  \& {Rogachevskii}, I. 2013, in IAU Symposium, Vol. 294, IAU Symposium, ed.
  A.~G. {Kosovichev}, E.~{de Gouveia Dal Pino}, \& Y.~{Yan}, 283--288

\bibitem[{{Kemel} {et~al.}(2012{\natexlab{a}}){Kemel}, {Brandenburg},
  {Kleeorin}, {Mitra}, \& {Rogachevskii}}]{Kemel12b}
{Kemel}, K., {Brandenburg}, A., {Kleeorin}, N., {Mitra}, D., \& {Rogachevskii},
  I. 2012{\natexlab{a}}, \solphys, 280, 321

\bibitem[{{Kemel} {et~al.}(2013){Kemel}, {Brandenburg}, {Kleeorin}, {Mitra}, \&
  {Rogachevskii}}]{Kemel13a}
{Kemel}, K., {Brandenburg}, A., {Kleeorin}, N., {Mitra}, D., \& {Rogachevskii},
  I. 2013, \solphys, 287, 293

\bibitem[{{Kemel} {et~al.}(2012{\natexlab{b}}){Kemel}, {Brandenburg},
  {Kleeorin}, \& {Rogachevskii}}]{Kemel12a}
{Kemel}, K., {Brandenburg}, A., {Kleeorin}, N., \& {Rogachevskii}, I.
  2012{\natexlab{b}}, Astron. Nachr., 333, 95

\bibitem[{{Kempf}(1910)}]{Kem10}
{Kempf}, P. 1910, Astron. Nachr., 185, 197

\bibitem[{{Kitchatinov} \& {Mazur}(2000)}]{KitchatinovMazur2000}
{Kitchatinov}, L.~L. \& {Mazur}, M.~V. 2000, \solphys, 191, 325

\bibitem[{{Kitiashvili} {et~al.}(2010){Kitiashvili}, {Kosovichev}, {Wray}, \&
  {Mansour}}]{KKWM2010}
{Kitiashvili}, I.~N., {Kosovichev}, A.~G., {Wray}, A.~A., \& {Mansour}, N.~N.
  2010, \apj, 719, 307

\bibitem[{Kleeorin {et~al.}(2003)Kleeorin, Kuzanyan, Moss, Rogachevskii,
  Sokoloff, \& Zhang}]{KKMRSZ03}
Kleeorin, N., Kuzanyan, K., Moss, D., {et~al.} 2003, Astron. \& Astrophys.,
  409, 1097

\bibitem[{{Kleeorin} {et~al.}(1993){Kleeorin}, {Mond}, \&
  {Rogachevskii}}]{KMR93}
{Kleeorin}, N., {Mond}, M., \& {Rogachevskii}, I. 1993, Phys. Fluids B, 5, 4128

\bibitem[{{Kleeorin} {et~al.}(1996){Kleeorin}, {Mond}, \&
  {Rogachevskii}}]{KMR96}
{Kleeorin}, N., {Mond}, M., \& {Rogachevskii}, I. 1996, \aap, 307, 293

\bibitem[{{Kleeorin} \& {Rogachevskii}(1994)}]{KR94}
{Kleeorin}, N. \& {Rogachevskii}, I. 1994, \pre, 50, 2716

\bibitem[{{Kleeorin} {et~al.}(1990){Kleeorin}, {Rogachevskii}, \&
  {Ruzmaikin}}]{KRR1990}
{Kleeorin}, N., {Rogachevskii}, I., \& {Ruzmaikin}, A. 1990, JETP, 70, 878

\bibitem[{{Kleeorin} {et~al.}(1989){Kleeorin}, {Rogachevskii}, \&
  {Ruzmaikin}}]{Kleeorin89}
{Kleeorin}, N.~I., {Rogachevskii}, I.~V., \& {Ruzmaikin}, A.~A. 1989, Sov.
  Astron. Lett., 15, 274

\bibitem[{{Losada} {et~al.}(2012){Losada}, {Brandenburg}, {Kleeorin}, {Mitra},
  \& {Rogachevskii}}]{LBKMR2012}
{Losada}, I.~R., {Brandenburg}, A., {Kleeorin}, N., {Mitra}, D., \&
  {Rogachevskii}, I. 2012, \aap, 548, A49

\bibitem[{{Losada} {et~al.}(2013){Losada}, {Brandenburg}, {Kleeorin}, \&
  {Rogachevskii}}]{LBKR2013a}
{Losada}, I.~R., {Brandenburg}, A., {Kleeorin}, N., \& {Rogachevskii}, I. 2013,
  \aap, 556, A83

\bibitem[{{Losada} {et~al.}(2014){Losada}, {Brandenburg}, {Kleeorin}, \&
  {Rogachevskii}}]{LBKR2014}
{Losada}, I.~R., {Brandenburg}, A., {Kleeorin}, N., \& {Rogachevskii}, I. 2014,
  \aap, 564, A2

\bibitem[{{Losada} {et~al.}(2017){Losada}, {Warnecke}, {Glogowski}, {Roth},
  {Brandenburg}, {Kleeorin}, \& {Rogachevskii}}]{LWGRBKR17}
{Losada}, I.~R., {Warnecke}, J., {Glogowski}, K., {et~al.} 2017, in IAU
  Symposium, Vol. 327, Fine Structure and Dynamics of the Solar Atmosphere, ed.
  S.~{Vargas Dom{\'{\i}}nguez}, A.~G. {Kosovichev}, P.~{Antolin}, \&
  L.~{Harra}, 46--59

\bibitem[{{Masada} \& {Sano}(2016)}]{MS16}
{Masada}, Y. \& {Sano}, T. 2016, \apjl, 822, L22

\bibitem[{{Mitra} {et~al.}(2014){Mitra}, {Brandenburg}, {Kleeorin}, \&
  {Rogachevskii}}]{MBKR14}
{Mitra}, D., {Brandenburg}, A., {Kleeorin}, N., \& {Rogachevskii}, I. 2014,
  \mnras, 445, 761

\bibitem[{{Parker}(1978)}]{Parker78}
{Parker}, E.~N. 1978, \apj, 221, 368

\bibitem[{{Perri} \& {Brandenburg}(2018)}]{PB18}
{Perri}, B. \& {Brandenburg}, A. 2018, \aap, 609, A99

\bibitem[{{Peter} {et~al.}(2015){Peter}, {Warnecke}, {Chitta}, \&
  {Cameron}}]{PWCC15}
{Peter}, H., {Warnecke}, J., {Chitta}, L.~P., \& {Cameron}, R.~H. 2015, \aap,
  584, A68

\bibitem[{{Pevtsov}(2012)}]{Pev12}
{Pevtsov}, A.~A. 2012, Astrophysics and Space Science Proceedings, 30, 83

\bibitem[{{Rempel} \& {Cheung}(2014)}]{rempel2014}
{Rempel}, M. \& {Cheung}, M.~C.~M. 2014, \apj, 785, 90

\bibitem[{{Rogachevskii} \& {Kleeorin}(2007)}]{RK07}
{Rogachevskii}, I. \& {Kleeorin}, N. 2007, \pre, 76, 056307

\bibitem[{{R{\"u}diger} {et~al.}(2011){R{\"u}diger}, {Kitchatinov}, \&
  {Brandenburg}}]{RKB11}
{R{\"u}diger}, G., {Kitchatinov}, L.~L., \& {Brandenburg}, A. 2011, \solphys,
  269, 3

\bibitem[{{R{\"u}diger} {et~al.}(2012){R{\"u}diger}, {Kitchatinov}, \&
  {Schultz}}]{RKS12}
{R{\"u}diger}, G., {Kitchatinov}, L.~L., \& {Schultz}, M. 2012, Astron. Nachr.,
  333, 84

\bibitem[{{Schmieder} {et~al.}(2014){Schmieder}, {Archontis}, \&
  {Pariat}}]{SAP14}
{Schmieder}, B., {Archontis}, V., \& {Pariat}, E. 2014, \ssr, 186, 227

\bibitem[{{Singh} {et~al.}(2016){Singh}, {Raichur}, \& {Brandenburg}}]{SRB16}
{Singh}, N.~K., {Raichur}, H., \& {Brandenburg}, A. 2016, \apj, 832, 120

\bibitem[{{Spruit}(1979)}]{Spruit79}
{Spruit}, H.~C. 1979, \solphys, 61, 363

\bibitem[{{Stein} \& {Nordlund}(2012)}]{SN2012}
{Stein}, R.~F. \& {Nordlund}, {\AA}. 2012, \apjl, 753, L13

\bibitem[{{Sturrock} {et~al.}(2015){Sturrock}, {Hood}, {Archontis}, \&
  {McNeill}}]{Stu15}
{Sturrock}, Z., {Hood}, A.~W., {Archontis}, V., \& {McNeill}, C.~M. 2015, \aap,
  582, A76

\bibitem[{{Sur} {et~al.}(2008){Sur}, {Brandenburg}, \& {Subramanian}}]{SBS08}
{Sur}, S., {Brandenburg}, A., \& {Subramanian}, K. 2008, \mnras, 385, L15

\bibitem[{{Tao} {et~al.}(1998){Tao}, {Weiss}, {Brownjohn}, \&
  {Proctor}}]{TaoWeiss98}
{Tao}, L., {Weiss}, N.~O., {Brownjohn}, D.~P., \& {Proctor}, M.~R.~E. 1998,
  \apjl, 496, L39

\bibitem[{{Warnecke} \& {Brandenburg}(2010)}]{WB2010}
{Warnecke}, J. \& {Brandenburg}, A. 2010, \aap, 523, A19

\bibitem[{{Warnecke} {et~al.}(2011){Warnecke}, {Brandenburg}, \&
  {Mitra}}]{WBM2011}
{Warnecke}, J., {Brandenburg}, A., \& {Mitra}, D. 2011, \aap, 534, A11

\bibitem[{{Warnecke} {et~al.}(2012{\natexlab{a}}){Warnecke}, {Brandenburg}, \&
  {Mitra}}]{WBM12}
{Warnecke}, J., {Brandenburg}, A., \& {Mitra}, D. 2012{\natexlab{a}}, JSWSC, 2,
  A11

\bibitem[{{Warnecke} {et~al.}(2017){Warnecke}, {Chen}, {Bingert}, \&
  {Peter}}]{WCBP17}
{Warnecke}, J., {Chen}, F., {Bingert}, S., \& {Peter}, H. 2017, \aap, 607, A53

\bibitem[{{Warnecke} {et~al.}(2016{\natexlab{a}}){Warnecke}, {K{\"a}pyl{\"a}},
  {K{\"a}pyl{\"a}}, \& {Brandenburg}}]{WKKB16}
{Warnecke}, J., {K{\"a}pyl{\"a}}, P.~J., {K{\"a}pyl{\"a}}, M.~J., \&
  {Brandenburg}, A. 2016{\natexlab{a}}, \aap, 596, A115

\bibitem[{{Warnecke} {et~al.}(2012{\natexlab{b}}){Warnecke}, {K{\"a}pyl{\"a}},
  {Mantere}, \& {Brandenburg}}]{WKMB12}
{Warnecke}, J., {K{\"a}pyl{\"a}}, P.~J., {Mantere}, M.~J., \& {Brandenburg}, A.
  2012{\natexlab{b}}, \solphys, 280, 299

\bibitem[{{Warnecke} {et~al.}(2013{\natexlab{a}}){Warnecke}, {K{\"a}pyl{\"a}},
  {Mantere}, \& {Brandenburg}}]{WKMB13}
{Warnecke}, J., {K{\"a}pyl{\"a}}, P.~J., {Mantere}, M.~J., \& {Brandenburg}, A.
  2013{\natexlab{a}}, \apj, 778, 141

\bibitem[{{Warnecke} {et~al.}(2013{\natexlab{b}}){Warnecke}, {Losada},
  {Brandenburg}, {Kleeorin}, \& {Rogachevskii}}]{WLBKR2013}
{Warnecke}, J., {Losada}, I.~R., {Brandenburg}, A., {Kleeorin}, N., \&
  {Rogachevskii}, I. 2013{\natexlab{b}}, \apjl, 777, L37

\bibitem[{{Warnecke} {et~al.}(2016{\natexlab{b}}){Warnecke}, {Losada},
  {Brandenburg}, {Kleeorin}, \& {Rogachevskii}}]{WLBKR2015}
{Warnecke}, J., {Losada}, I.~R., {Brandenburg}, A., {Kleeorin}, N., \&
  {Rogachevskii}, I. 2016{\natexlab{b}}, \aap, 589, A125

\bibitem[{{Yousef} {et~al.}(2003){Yousef}, {Brandenburg}, \&
  {R{\"u}diger}}]{YBR03}
{Yousef}, T.~A., {Brandenburg}, A., \& {R{\"u}diger}, G. 2003, \aap, 411, 321

\bibitem[{{Zhang} \& {Brandenburg}(2018)}]{ZB18}
{Zhang}, H. \& {Brandenburg}, A. 2018, \apjl, 862, L17

\end{thebibliography}

% $Id: paper.tex,v 1.417 2018/08/28 09:36:28 joern Exp $

\documentclass{aa}
%\documentclass[referee]{aa}
\usepackage{graphicx,color}
\usepackage{bm,natbib,url,wasysym}
\usepackage[varg]{txfonts}
\bibpunct{(}{)}{;}{a}{}{,} % to follow the A&A style
\graphicspath{{./figs/}{./png/}}

%\input macros
%\def\blue{\textcolor{blue}}
%\def\blue{\textcolor{black}}
%\def\red{\textcolor{red}}

%|||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||
%             Customized Commands
%|||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||||
%  mathematical abbreviations
%  =========================

% math defs
\newcommand{\EQ}{\begin{equation}}
\newcommand{\EN}{\end{equation}}
\newcommand{\EQA}{\begin{eqnarray}}
\newcommand{\ENA}{\end{eqnarray}}
\newcommand{\eq}[1]{(\ref{#1})}
\newcommand{\EEq}[1]{Equation~(\ref{#1})}
\newcommand{\Eq}[1]{Eq.~(\ref{#1})}
\newcommand{\Eqs}[2]{Eqs.~(\ref{#1}) and~(\ref{#2})}
\newcommand{\Eqss}[2]{Eqs.~(\ref{#1})--(\ref{#2})}
\newcommand{\eqs}[2]{(\ref{#1}) and~(\ref{#2})}
\newcommand{\App}[1]{Appendix~\ref{#1}}
\newcommand{\Sec}[1]{Sect.~\ref{#1}}
\newcommand{\Secs}[2]{Sects.~\ref{#1} and \ref{#2}}
%
%\newcommand{\FFig}[1]{Figure~\ref{#1}}
%\newcommand{\Fig}[1]{Fig.~\ref{#1}}
%\newcommand{\Figs}[2]{Figs.~\ref{#1} and \ref{#2}}
%\newcommand{\Figss}[2]{Figs.~\ref{#1}--\ref{#2}}
%\newcommand{\Tab}[1]{Table~\ref{#1}}
%\newcommand{\Tabs}[2]{Tables~\ref{#1} and \ref{#2}}

%AB: replaced those macros by those in color (during working stage)
\newcommand{\FFig}[1]{\textcolor{red}{Figure~\ref{#1}}}
\newcommand{\Fig}[1]{\textcolor{red}{Fig.~\ref{#1}}}
\newcommand{\Figs}[2]{\textcolor{red}{Figs.~\ref{#1} and \ref{#2}}}
\newcommand{\Figss}[2]{\textcolor{red}{Figs.~\ref{#1}--\ref{#2}}}
\newcommand{\Figp}[2]{\textcolor{red}{Fig.~\ref{#1}({#2})}}
\newcommand{\Figsp}[3]{\textcolor{red}{Figs.~\ref{#1}({#2}) and ({#3})}}
\newcommand{\Figssp}[3]{\textcolor{red}{Figs.~\ref{#1}({#2})--({#3})}}
\newcommand{\Tab}[1]{\textcolor{magenta}{Table~\ref{#1}}}
%\newcommand{\Tab}[1]{\textcolor{red}{Table~\ref{#1}}}
\newcommand{\Tabs}[2]{\textcolor{red}{Tables~\ref{#1} and \ref{#2}}}
\newcommand{\Tabss}[2]{\textcolor{red}{Tables~\ref{#1}--\ref{#2}}}
%
\newcommand{\bra}[1]{\langle #1\rangle}
\newcommand{\bbra}[1]{\left\langle #1\right\rangle}
\newcommand{\mean}[1]{\overline #1}
\newcommand{\meanf}[1]{\overline #1{}^{\rm fil}}
\newcommand{\meanrho}{\overline{\rho}}
\newcommand{\meanPhi}{\overline{\Phi}}
\newcommand{\tildeFFFF}{\tilde{\mbox{\boldmath ${\cal F}$}}{}}{}
\newcommand{\hatFFFF}{\hat{\mbox{\boldmath ${\cal F}$}}{}}{}
\newcommand{\meanFFFF}{\overline{\mbox{\boldmath ${\cal F}$}}{}}{}
\newcommand{\meanemf}{\overline{\cal E} {}}
\newcommand{\meanAAAA}{\overline{\mbox{\boldmath ${\mathsf A}$}} {}}
\newcommand{\meanSSSS}{\overline{\mbox{\boldmath ${\mathsf S}$}} {}}
\newcommand{\meanAAA}{\overline{\mathsf{A}}}
\newcommand{\meanSSS}{\overline{\mathsf{S}}}
\newcommand{\meanuu}{\overline{\mbox{\boldmath $u$}}{}}{}
\newcommand{\meanoo}{\overline{\mbox{\boldmath $\omega$}}{}}{}
\newcommand{\meanEMF}{\overline{\mbox{\boldmath ${\cal E}$}}{}}{}
\newcommand{\meanEEEE}{\overline{\mbox{\boldmath ${\cal E}$}}{}}{}
\newcommand{\meanuxB}{\overline{\mbox{\boldmath $\delta u\times \delta B$}}{}}{}
\newcommand{\meanJB}{\overline{\mbox{\boldmath $J\cdot B$}}{}}{}
\newcommand{\meanAB}{\overline{\mbox{\boldmath $A\cdot B$}}{}}{}
\newcommand{\meanjb}{\overline{\mbox{\boldmath $j\cdot b$}}{}}{}
\newcommand{\meanAA}{\overline{\mbox{\boldmath $A$}}{}}{}
\newcommand{\meanBB}{\overline{\mbox{\boldmath $B$}}{}}{}
\newcommand{\meanEE}{\overline{\mbox{\boldmath $E$}}{}}{}
\newcommand{\meanFF}{\overline{\mbox{\boldmath $F$}}{}}{}
\newcommand{\meanFFf}{\overline{\mbox{\boldmath $F$}}_{\rm f}{}}{}
\newcommand{\meanFFm}{\overline{\mbox{\boldmath $F$}}_{\rm m}{}}{}
\newcommand{\hatFF}{\hat{\mbox{\boldmath $F$}}{}}{}
\newcommand{\meanGG}{\overline{\mbox{\boldmath $G$}}{}}{}
\newcommand{\meanHH}{\overline{\mbox{\boldmath $H$}}{}}{}
\newcommand{\tildeGG}{\tilde{\mbox{\boldmath $G$}}{}}{}
\newcommand{\meanJJ}{\overline{\mbox{\boldmath $J$}}{}}{}
\newcommand{\meanKK}{\overline{\mbox{\boldmath $K$}}{}}{}
\newcommand{\meanUU}{\overline{\mbox{\boldmath $U$}}{}}{}
\newcommand{\meanVV}{\overline{\bm{V}}}
\newcommand{\meanWW}{\overline{\mbox{\boldmath $W$}}{}}{}
\newcommand{\hatWW}{\hat{\mbox{\boldmath $W$}}{}}{}
\newcommand{\meanQQ}{\overline{\mbox{\boldmath $Q$}}{}}{}
\newcommand{\meanA}{\overline{A}}
\newcommand{\meanB}{\overline{B}}
\newcommand{\meanE}{\overline{E}}
\newcommand{\meanK}{\overline{K}}
\newcommand{\meanb}{\tilde{b}}
\newcommand{\meanj}{\tilde{j}}
\newcommand{\meanC}{\overline{C}}
\newcommand{\meanD}{\overline{D}}
\newcommand{\meanF}{\overline{F}}
\newcommand{\meanFm}{\overline{F}_{\rm m}}
\newcommand{\meanFf}{\overline{F}_{\rm f}}
\newcommand{\meanh}{\overline{h}}
\newcommand{\meanhm}{\overline{h}_{\rm m}}
\newcommand{\meanhf}{\overline{h}_{\rm f}}
\newcommand{\meanG}{\overline{G}}
\newcommand{\meanGGG}{\overline{\cal G}}
\newcommand{\meanH}{\overline{H}}
\newcommand{\meanU}{\overline{U}}
\newcommand{\meanR}{\overline{R}}
\newcommand{\meanJ}{\overline{J}}
\newcommand{\means}{\overline{s}}
\newcommand{\meanp}{\overline{p}}
\newcommand{\meanS}{\overline{S}}
\newcommand{\meanT}{\overline{T}}
\newcommand{\meanW}{\overline{W}}
\newcommand{\meanQ}{\overline{Q}}
\newcommand{\meanEEE}{\overline{\cal E}}
\newcommand{\meanFFF}{\overline{\cal F}}
\newcommand{\meanHHH}{\overline{\cal H}}
\newcommand{\meanRRR}{\overline{\cal R}}
%\newcommand{\hatAA}{\hat{\mbox{\boldmath $A$}}{}}{}
%\newcommand{\hatBB}{\hat{\mbox{\boldmath $B$}}{}}{}
\newcommand{\hatg}{\hat{g}}
\newcommand{\hatk}{\hat{k}}
\newcommand{\hatkk}{\hat{\bm{k}}}
\newcommand{\hatAA}{\hat{\bm{A}}}
\newcommand{\hatBB}{\hat{\vec{B}}}
\newcommand{\hatJJ}{\hat{\mbox{\boldmath $J$}}{}}{}
%\newcommand{\hatOO}{\hat{\mbox{\boldmath $\Omega$}} {}}
\newcommand{\hatOO}{\hat{\bm{\Omega}}}
\newcommand{\hatEMF}{\hat{\mbox{\boldmath ${\cal E}$}}{}}{}
\newcommand{\emf}{{\cal E}}{}
\newcommand{\hatA}{\hat{A}}
\newcommand{\hatB}{\hat{B}}
\newcommand{\hatJ}{\hat{J}}
\newcommand{\hatU}{\hat{U}}
\newcommand{\alphaK}{\alpha_{\it K}}
\newcommand{\alphaM}{\alpha_{\it M}}
\newcommand{\alphaSS}{\alpha_{\it SS}}
%
% tilde
%
\newcommand{\tildeB}{\tilde{B}}
\newcommand{\tildeJ}{\tilde{J}}
\newcommand{\tildeemf}{\tilde{\cal E}}
\newcommand{\teps}{\tilde{\epsilon} {}}
\newcommand{\Oh}{\hat{\Omega}}
\newcommand{\zh}{\hat{z}}
\newcommand{\PC}{{\sc Pencil Code}~}
\newcommand{\PCn}{{\sc Pencil Code}}
%
%  unit vectors
%
\newcommand{\pphi}{\hat{\bm{\phi}}}
\newcommand{\ppom}{\bm{\hat{\varpi}}}
%\newcommand{\ppom}{\hat{\bm{\varpi}}}
\newcommand{\eee}{\hat{\mbox{\boldmath $e$}} {}}
\newcommand{\nnn}{\hat{\mbox{\boldmath $n$}} {}}
\newcommand{\rrr}{\hat{\mbox{\boldmath $r$}} {}}
\newcommand{\vvv}{\hat{\mbox{\boldmath $v$}} {}}
\newcommand{\xxx}{\hat{\mbox{\boldmath $x$}} {}}
\newcommand{\yyy}{\hat{\mbox{\boldmath $y$}} {}}
\newcommand{\zzz}{\hat{\mbox{\boldmath $z$}} {}}
\newcommand{\ttt}{\hat{\mbox{\boldmath $\theta$}} {}}
\newcommand{\OOO}{\hat{\mbox{\boldmath $\Omega$}} {}}
\newcommand{\ooo}{\hat{\mbox{\boldmath $\omega$}} {}}
%\newcommand{\ppom}{\hat{\mbox{\boldmath $\varpi$}} {}}
\newcommand{\BBBB}{\hat{\mbox{\boldmath $B$}} {}}
\newcommand{\meanBBhat}{\hat{\overline{\bm{B}}}}
\newcommand{\meanJJhat}{\hat{\overline{\bm{J}}}}
\newcommand{\meanBhat}{\hat{\overline{B}}}
\newcommand{\meanJhat}{\hat{\overline{J}}}
\newcommand{\Bhat}{\hat{B}}
\newcommand{\Bh}{\hat{B}}
%
%  vectors
%
\newcommand{\nullvector}{{\bf0}}
\newcommand{\gggg}{\mbox{\boldmath $g$} {}}
\newcommand{\ddd}{\mbox{\boldmath $d$} {}}
\newcommand{\rr}{\mbox{\boldmath $r$} {}}
\newcommand{\yy}{\mbox{\boldmath $y$} {}}
\newcommand{\zz}{\mbox{\boldmath $z$} {}}
\newcommand{\vv}{\mbox{\boldmath $v$} {}}
\newcommand{\ww}{\mbox{\boldmath $w$} {}}
\newcommand{\mm}{\mbox{\boldmath $m$} {}}
\newcommand{\PP}{\mbox{\boldmath $P$} {}}
\newcommand{\bp}{\mbox{\boldmath $p$} {}}
\newcommand{\II}{\mbox{\boldmath $I$} {}}
\newcommand{\RR}{\mbox{\boldmath $R$} {}}
\newcommand{\kk}{\bm{k}}
\newcommand{\pp}{\bm{p}}
\newcommand{\qq}{\bm{q}}
\newcommand{\xx}{\bm{x}}
\newcommand{\XX}{\bm{X}}
\newcommand{\bb}{\bm{b}}
\newcommand{\BB}{\bm{B}}
\newcommand{\HH}{\bm{H}}
\newcommand{\MM}{\bm{M}}
\newcommand{\KK}{\mbox{\boldmath $K$} {}}
\newcommand{\uu}{\mbox{\boldmath $u$} {}}
\newcommand{\UU}{\mbox{\boldmath $U$} {}}
\newcommand{\sss}{\mbox{\boldmath $s$} {}}
%\newcommand{\aaa}{\mbox{\boldmath $a$} {}}
%\newcommand{\bb}{\mbox{\boldmath $b$} {}}
%\newcommand{\BB}{\mbox{\boldmath $B$} {}}
\newcommand{\EE}{\mbox{\boldmath $E$} {}}
\newcommand{\jj}{\mbox{\boldmath $j$} {}}
\newcommand{\JJ}{\mbox{\boldmath $J$} {}}
\newcommand{\SSS}{\mbox{\boldmath $S$} {}}
\newcommand{\AAA}{\mbox{\boldmath $A$} {}}
\newcommand{\aaaa}{\mbox{\boldmath $a$} {}}
\newcommand{\ee}{\mbox{\boldmath $e$} {}}
\newcommand{\nn}{\mbox{\boldmath $n$} {}}
\newcommand{\ff}{\mbox{\boldmath $f$} {}}
\newcommand{\hh}{\mbox{\boldmath $h$} {}}
\newcommand{\FF}{\mbox{\boldmath $F$} {}}
\newcommand{\EEE}{\mbox{\boldmath ${\cal E}$} {}}
\newcommand{\FFF}{\mbox{\boldmath ${\cal F}$} {}}
\newcommand{\TT}{{\bm{T}}}
\newcommand{\GG}{\mbox{\boldmath $G$} {}}
\newcommand{\WW}{\mbox{\boldmath $W$} {}}
\newcommand{\QQ}{\mbox{\boldmath $Q$} {}}
\newcommand{\grav}{\mbox{\boldmath $g$} {}}
%\newcommand{\ggg}{\mbox{\boldmath $g$} {}}
\newcommand{\nab}{\mbox{\boldmath $\nabla$} {}}
%\newcommand{\OO}{\mbox{\boldmath $\Omega$} {}}
\newcommand{\OO}{\bm{\Omega}}
\newcommand{\ppsi}{\mbox{\boldmath $\psi$} {}}
\newcommand{\oo}{\mbox{\boldmath $\omega$} {}}
\newcommand{\pom}{\mbox{\boldmath $\varpi$} {}}
\newcommand{\ttau}{\mbox{\boldmath $\tau$} {}}
\newcommand{\LL}{\mbox{\boldmath $\Lambda$} {}}
\newcommand{\mmu}{\mbox{\boldmath $\mu$} {}}
\newcommand{\xxi}{\mbox{\boldmath $\xi$} {}}
\newcommand{\ddelta}{\mbox{\boldmath $\delta$} {}}
\newcommand{\ggamma}{\mbox{\boldmath $\gamma$} {}}
\newcommand{\kkappa}{\mbox{\boldmath $\kappa$} {}}
\newcommand{\llambda}{\mbox{\boldmath $\lambda$} {}}
\newcommand{\pomega}{\mbox{\boldmath $\varpi$} {}}
\newcommand{\PPsi}{\mbox{\boldmath $\Psi$} {}}
%
%  tensors
%
\newcommand{\DDDD}{\mbox{\boldmath ${\sf D}$} {}}
\newcommand{\IIII}{\mbox{\boldmath ${\sf I}$} {}}
\newcommand{\LLLL}{\mbox{\boldmath ${\sf L}$} {}}
\newcommand{\MMMM}{\mbox{\boldmath ${\sf M}$} {}}
\newcommand{\NNNN}{\mbox{\boldmath ${\sf N}$} {}}
\newcommand{\PPPP}{\mbox{\boldmath ${\sf P}$} {}}
\newcommand{\QQQQ}{\mbox{\boldmath ${\sf Q}$} {}}
\newcommand{\RRRR}{\mbox{\boldmath ${\sf R}$} {}}
\newcommand{\SSSS}{\mbox{\boldmath ${\sf S}$} {}}
\newcommand{\BBBBB}{\mbox{\boldmath ${\sf B}$} {}}
\newcommand{\tAAAA}{\tilde{\mbox{\boldmath ${\sf A}$}} {}}
\newcommand{\tDDDD}{\tilde{\mbox{\boldmath ${\sf D}$}} {}}
\newcommand{\tRRRR}{\tilde{\mbox{\boldmath ${\sf R}$}} {}}
\newcommand{\tQQQQ}{\tilde{\mbox{\boldmath ${\sf Q}$}} {}}
%
\newcommand{\AAAA}{\mbox{\boldmath ${\cal A}$} {}}
\newcommand{\BBB}{\mbox{\boldmath ${\cal B}$} {}}
\newcommand{\EMF}{\mbox{\boldmath ${\cal E}$} {}}
\newcommand{\GGG}{\mbox{\boldmath ${\cal G}$} {}}
\newcommand{\HHH}{\mbox{\boldmath ${\cal H}$} {}}
%\newcommand{\QQQ}{\mbox{\boldmath ${\cal Q}$} {}}
%
\newcommand{\GGGG}{{\bf G} {}}
%
%  operators, subscripts, etc  (roman)
%
\newcommand{\ii}{{\rm i}}
\newcommand{\erf}{{\rm erf}}
\newcommand{\grad}{{\rm grad} \, {}}
\newcommand{\curl}{{\rm curl} \, {}}
\newcommand{\dive}{{\rm div}  \, {}}
\newcommand{\Dive}{{\rm Div}  \, {}}
\newcommand{\diag}{{\rm diag}  \, {}}
\newcommand{\sgn}{{\rm sgn}  \, {}}
\newcommand{\meanDD}{{\overline{\rm D}} {}}
\newcommand{\DD}{{\rm D} {}}
\newcommand{\DDD}{{\cal D} {}}
\newcommand{\QQQ}{{\cal Q}}
\newcommand{\dd}{{\rm d} {}}
\newcommand{\dV}{\,{\rm d}V {}}
\newcommand{\dS}{\,{\rm d}{{\bm{S}}} {}}
\newcommand{\const}{{\rm const}  {}}
\newcommand{\CR}{{\rm CR}}
\def\degr{\hbox{$^\circ$}}
\def\la{\mathrel{\mathchoice {\vcenter{\offinterlineskip\halign{\hfil
$\displaystyle##$\hfil\cr<\cr\sim\cr}}}
{\vcenter{\offinterlineskip\halign{\hfil$\textstyle##$\hfil\cr<\cr\sim\cr}}}
{\vcenter{\offinterlineskip\halign{\hfil$\scriptstyle##$\hfil\cr<\cr\sim\cr}}}
{\vcenter{\offinterlineskip\halign{\hfil$\scriptscriptstyle##$\hfil\cr<\cr\sim\cr}}}}}
\def\ga{\mathrel{\mathchoice {\vcenter{\offinterlineskip\halign{\hfil
$\displaystyle##$\hfil\cr>\cr\sim\cr}}}
{\vcenter{\offinterlineskip\halign{\hfil$\textstyle##$\hfil\cr>\cr\sim\cr}}}
{\vcenter{\offinterlineskip\halign{\hfil$\scriptstyle##$\hfil\cr>\cr\sim\cr}}}
{\vcenter{\offinterlineskip\halign{\hfil$\scriptscriptstyle##$\hfil\cr>\cr\sim\cr}}}}}
%
%  numbers
%
\def\Ta{\mbox{\rm Ta}}
\def\Ra{\mbox{\rm Ra}}
\def\Ma{\mbox{\rm Ma}}
\def\Co{\mbox{\rm Co}}
\def\Ro{\mbox{\rm Ro}}
\def\CoM{\mbox{\rm Co}_{\rm M}}
\def\Sh{\mbox{\rm Sh}}
\def\St{\mbox{\rm St}}
\def\Roo{\mbox{\rm Ro}^{-1}}
\def\Rooo{\mbox{\rm Ro}^{-2}}
\def\Pra{\mbox{\rm Pr}}
\def\Pran{\mbox{\rm Pr}}
\def\Sc{\mbox{\rm Sc}}
\def\Tr{\mbox{\rm Tr}}
\def\Da{\mbox{\rm Da}}
\def\Ka{\mbox{\rm Ka}}
\def\Pm{\mbox{\rm Pr}_M}
\def\Rm{\mbox{\rm Re}_M}
\def\Tat{\mbox{\rm Ta}_{\rm t}}
%\def\Rm{R_{\rm m}}
\def\Dt{D_{\rm t}}
\def\DT{D_{\rm T}}
%\def\Rmc{\mbox{\rm Re}_{M,{\rm crit}}}
%\def\Pm{P_{\rm m}}
%\def\Rm{R_{\rm m}}
\def\Rmc{R_{\rm m,{\rm crit}}}
\def\Rey{\mbox{\rm Re}}
\def\Pe{\mbox{\rm Pe}}
\def\Co{\mbox{\rm Co}}
\def\Lu{\mbox{\rm Lu}}
\def\tauc{\tau_{\rm c}}
\def\taueff{\tau_{\rm eff}}
\def\alpK{\alpha_{\rm K}}
\def\alpM{\alpha_{\rm M}}
\def\EEK{{\cal E}_{\rm K}}
\def\EEM{{\cal E}_{\rm M}}
\def\EK{E_{\rm K}}
\def\EM{E_{\rm M}}
\def\ET{E_{\rm T}}
\def\WL{W_{\rm L}}
\def\WC{W_{\rm C}}
\def\csz{c_{\rm s0}}
\def\Teff{T_{\rm eff}}
\def\cp{c_{\rm p}}
\def\cv{c_{\rm v}}
\def\Rs{R_{\odot}}
\def\cs{c_{\rm s}}
\def\Hpz{H_{\rm p0}}
\def\Hrz{H_{\rho 0}}
\def\qpz{q_{\rm p0}}
\def\qp{q_{\rm p}}
\def\ellK{\ell_{\rm K}}
\def\betap{\beta_{\rm p}}
\def\betamin{\beta_{\min}}
\def\betastar{\beta_{\star}}
\def\Peff{{\cal P}_{\rm eff}}
\def\Peffmin{{\cal P}_{\rm eff}^{\rm min}}
%\def\Pmin{{\cal P}_{\rm min}}
%AB: modified
\def\Pmin{{\cal P}_{\rm eff}^{\rm min}}
\def\pturbz{p_{\rm turb}^{(0)}}
\def\pturb{p_{\rm turb}}
\def\peff{p_{\rm eff}}
\def\qs{q_{\rm s}}
\def\qe{q_{\rm e}}
\def\qg{q_{\rm g}}
\def\qa{q_{\rm a}}
\def\pt{p_{\rm t}}
\def\ptz{p_{\rm t0}}
\def\vA{v_{\rm A}}
\def\vAz{v_{\rm A0}}
\def\hf{h_{\rm f}}
\def\hm{h_{\rm m}}
\def\hrms{h_{\rm rms}}
\def\kmean{k_{\rm m}}
\def\km{k_{\rm m}}
\def\lf{l_{\rm f}}
\def\kone{k_1}
\def\kd{k_{\rm d}}
\def\kf{k_{\rm f}}
\def\HP{H_p}
%AB: italics for ApJ
%\def\kf{k_{\it f}}
\def\HM{H_{\rm M}}
\def\EM{E_{\rm M}}
\def\kappaf{\kappa_{\rm f}}
\def\epsf{\epsilon_{\rm f}}
\def\epsK{\epsilon_{\it K}}
\def\epsM{\epsilon_{\it M}}
\def\epsC{\epsilon_{\it C}}
\def\epsT{\epsilon_{\it T}}
\def\xid{\xi_{\rm D}}
\def\kK{k_{\it K}}
\def\kM{k_{\it M}}
\def\kmax{k_{\rm max}}
\def\kom{k_{\omega}}
\def\Ff{F_{\rm f}}
\def\Fm{F_{\rm m}}
\def\Hf{H_{\rm f}}
\def\Hm{H_{\rm m}}
\def\vArms{v_{\rm A,rms}}
\def\Brms{B_{\rm rms}}
\def\Jrms{J_{\rm rms}}
\def\Urms{U_{\rm rms}}
\def\urms{u_{\rm rms}}
\def\urmsz{u_{\rm rms0}}
\def\wrms{w_{\rm rms}}
\def\orms{\omega_{\rm rms}}
\def\brms{b_{\rm rms}}
\def\uref{u_{\rm ref}}
\def\yN{y_{\rm N}}
\def\yL{y_{\rm L}}
\def\kappaBB{\kappa_{\rm BB}}
\def\kappaB{\kappa_{\rm B}}
\def\kappaOO{\kappa_{\Omega\Omega}}
\def\kappaO{\kappa_{\Omega}}
\def\kappah{\kappa_{\rm h}}
\def\kappat{\kappa_{\rm t}}
\def\kappaT{\kappa_{\rm T}}
\def\kappatz{\kappa_{\rm t0}}
\def\mut{\mu_{\rm t}}
\def\nut{\nu_{\rm t}}
\def\nuM{\nu_{\rm M}}
\def\nuT{\nu_{\rm T}}
\def\calKt{{\cal K}_{\rm t}}
\def\calKtz{{\cal K}_{\rm t0}}
\def\chit{\chi_{\rm t}}
\def\chitz{\chi_{\rm t0}}
\def\etat{\eta_{\rm t}}
%AB: for ApJ they have italics t here.
%\def\etat{\eta_{\it t}}
\def\etatz{\eta_{\rm t0}}
\def\etaTz{\eta_{\rm T0}}
\def\nutz{\nu_{\rm t0}}
\def\etaT{\eta_{\rm T}}
\def\mut{\mu_{\rm t}}
\def\muT{\mu_{\rm T}}
\def\uT{\mu{\rm T}}
\def\BBeq{|\BB|/B_{\rm eq}}
\def\Beq{B_{\rm eq}}
\def\Beqz{B_{\rm eq0}}
\def\Bfm{\overline{B}{}_z^{\rm fil\,max}}
\def\tautd{\tau_{\rm td}}
\def\tauto{\tau_{\rm to}}
\def\Gr{\mbox{\rm Gr}}
\def\tautdm{\tau_{\rm td}^{\rm max}}
\def\csq{c_{\rm s}^2}
%
\def\Bfm{\overline{B}{}_z^{\rm fil\,max}}
%
\newcommand{\ea}{{\rm et al.\ }}
\newcommand{\eaa}{{\rm et al.\ }}
\def\half{{\textstyle{1\over2}}}
\def\threehalf{{\textstyle{3\over2}}}
\def\threequarter{{\textstyle{3\over4}}}
\def\sevenhalf{{\textstyle{7\over2}}}
\def\onethird{{\textstyle{1\over3}}}
\def\onesixth{{\textstyle{1\over6}}}
\def\twothird{{\textstyle{2\over3}}}
\def\onenineth{{\textstyle{1\over9}}}
\def\fourthird{{\textstyle{4\over3}}}
\def\quarter{{\textstyle{1\over4}}}
%
\newcommand{\W}{\,{\rm W}}
\newcommand{\V}{\,{\rm V}}
\newcommand{\kV}{\,{\rm kV}}
\newcommand{\T}{\,{\rm T}}
\newcommand{\G}{\,{\rm G}}
\newcommand{\Hz}{\,{\rm Hz}}
\newcommand{\nHz}{\,{\rm nHz}}
\newcommand{\kHz}{\,{\rm kHz}}
\newcommand{\kG}{\,{\rm kG}}
\newcommand{\K}{\,{\rm K}}
\newcommand{\g}{\,{\rm g}}
\newcommand{\s}{\,{\rm s}}
\newcommand{\ms}{\,{\rm ms}}
\newcommand{\mpers}{\,{\rm m/s}}
\newcommand{\ks}{\,{\rm ks}}
\newcommand{\cm}{\,{\rm cm}}
\newcommand{\mV}{\,{\rm mV}}
\newcommand{\cmcube}{\,{\rm cm^{-3}}}
\newcommand{\m}{\,{\rm m}}
%\newcommand{\km}{\,{\rm km}}
\newcommand{\KM}{\,{\rm km}}
\newcommand{\msec}{\,{\rm ms}}
\newcommand{\cms}{\,{\rm cm/s}}
\newcommand{\kms}{\,{\rm km/s}}
\newcommand{\gpercc}{\,{\rm g/cm}^3}
\newcommand{\kg}{\,{\rm kg}}
\newcommand{\ug}{\,\mu{\rm g}}
\newcommand{\kW}{\,{\rm kW}}
\newcommand{\MW}{\,{\rm MW}}
\newcommand{\Mm}{\,{\rm Mm}}
\newcommand{\Mx}{\,{\rm Mx}}
\newcommand{\pc}{\,{\rm pc}}
\newcommand{\kpc}{\,{\rm kpc}}
\newcommand{\yr}{\,{\rm yr}}
\newcommand{\Myr}{\,{\rm Myr}}
\newcommand{\Gyr}{\,{\rm Gyr}}
\newcommand{\erg}{\,{\rm erg}}
\newcommand{\mol}{\,{\rm mol}}
\newcommand{\dyn}{\,{\rm dyn}}
\newcommand{\J}{\,{\rm J}}
%\newcommand{\RM}{\,{\rm RM}}
%\newcommand{\EM}{\,{\rm EM}}
\newcommand{\AU}{\,{\rm AU}}
\newcommand{\A}{\,{\rm A}}
%\newcommand{\kA}{\,{\rm kA}}
%\newcommand{\etal}{et al.}
\newcommand{\h}{\,{\rm h}}
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%  journals
%
\newcommand{\yastroph}[2]{ #1, astro-ph/#2}
\newcommand{\ycsf}[3]{ #1, {Chaos, Solitons \& Fractals,} {#2}, #3}
\newcommand{\yepl}[3]{ #1, {Europhys.\ Lett.,} {#2}, #3}
\newcommand{\yaj}[3]{ #1, {AJ,} {#2}, #3}
\newcommand{\yjgr}[3]{ #1, {J.\ Geophys.\ Res.,} {#2}, #3}
\newcommand{\ysol}[3]{ #1, {Sol.\ Phys.,} {#2}, #3}
\newcommand{\ymonber}[3]{ #1, {Monats.\ Dt.\ Akad.\ Wiss.,} {#2}, #3}
\newcommand{\yapj}[3]{ #1, {ApJ,} {#2}, #3}
\newcommand{\ypasp}[3]{ #1, {PASP,} {#2}, #3}
\newcommand{\yapjl}[3]{ #1, {ApJ,} {#2}, #3}
\newcommand{\yapjs}[3]{ #1, {ApJS,} {#2}, #3}
\newcommand{\yija}[3]{ #1, {Int.\ J.\ Astrobiol.,} {#2}, #3}
\newcommand{\yan}[3]{ #1, {Astron.\ Nachr.,} {#2}, #3}
\newcommand{\yzfa}[3]{ #1, {Z.\ f.\ Ap.,} {#2}, #3}
\newcommand{\ymhdn}[3]{ #1, {Magnetohydrodyn.} {#2}, #3}
\newcommand{\yana}[3]{ #1, {A\&A,} {#2}, #3}
\newcommand{\yanaN}[4]{:~#1, ``#4,'' {\em Astron.\ Astrophys.\ }{\bf #2}, #3}
\newcommand{\yanas}[3]{ #1, {A\&AS,} {#2}, #3}
\newcommand{\yanar}[3]{ #1, {A\&A Rev.,} {#2}, #3}
\newcommand{\yass}[3]{ #1, {Ap\&SS,} {#2}, #3}
\newcommand{\ygafd}[3]{ #1, {Geophys.\ Astrophys.\ Fluid Dyn.,} {#2}, #3}
\newcommand{\ygrl}[3]{ #1, {Geophys.\ Res.\ Lett.,} {#2}, #3}
\newcommand{\ypasj}[3]{ #1, {Publ.\ Astron.\ Soc.\ Japan,} {#2}, #3}
\newcommand{\yjfm}[3]{ #1, {J.\ Fluid Mech.,} {#2}, #3}
\newcommand{\ypepi}[3]{ #1, {Phys.\ Earth Planet.\ Int.,} {#2}, #3}
\newcommand{\ypf}[3]{ #1, {Phys.\ Fluids,} {#2}, #3}
\newcommand{\ypfb}[3]{ #1, {Phys.\ Fluids B,} {#2}, #3}
\newcommand{\ypp}[3]{ #1, {Phys.\ Plasmas,} {#2}, #3}
\newcommand{\ysov}[3]{ #1, {Sov.\ Astron.,} {#2}, #3}
\newcommand{\ysovl}[3]{ #1, {Sov.\ Astron.\ Lett.,} {#2}, #3}
\newcommand{\yjetp}[3]{ #1, {Sov.\ Phys.\ JETP,} {#2}, #3}
\newcommand{\yphy}[3]{ #1, {Physica,} {#2}, #3}
\newcommand{\yaraa}[3]{ #1, {ARA\&A,} {#2}, #3}
\newcommand{\yanf}[3]{ #1, {Ann. Rev. Fluid Mech.,} {#2}, #3}
\newcommand{\yrpp}[3]{ #1, {Rep.\ Prog.\ Phys.,} {#2}, #3}
\newcommand{\yprs}[3]{ #1, {Proc.\ Roy.\ Soc.\ Lond.,} {#2}, #3}
\newcommand{\yprt}[3]{ #1, {Phys.\ Rep.,} {#2}, #3}
\newcommand{\yprl}[3]{ #1, {Phys.\ Rev.\ Lett.,} {#2}, #3}
\newcommand{\yphl}[3]{ #1, {Phys.\ Lett.,} {#2}, #3}
\newcommand{\yptrs}[3]{ #1, {Phil.\ Trans.\ R.\ Soc.,} {#2}, #3}
\newcommand{\yptrsa}[3]{ #1, {Phil. Trans. R. Soc. A,} {#2}, #3}
\newcommand{\ymn}[3]{ #1, {MNRAS,} {#2}, #3}
\newcommand{\ynat}[3]{ #1, {Nature,} {#2}, #3}
\newcommand{\ysci}[3]{ #1, {Science,} {#2}, #3}
\newcommand{\ysph}[3]{ #1, {Solar Phys.,} {#2}, #3}
\newcommand{\psph}[2]{ #1, {Solar Phys.,} {#2}}
\newcommand{\yrmp}[3]{ #1, {Rev.\ Mod.\ Phys.} {#2}, #3}
\newcommand{\ypr}[3]{ #1, {Phys.\ Rev.,} {#2}, #3}
\newcommand{\yprd}[3]{ #1, {Phys.\ Rev.\ D,} {#2}, #3}
\newcommand{\ypre}[3]{ #1, {Phys.\ Rev.\ E,} {#2}, #3}
\newcommand{\ypnas}[3]{ #1, {Proc.\ Nat.\ Acad.\ Sci.,} {#2}, #3}
\newcommand{\yssr}[3]{ #1, {Spa.\ Sci.\ Rev.,} {#2}, #3}
\newcommand{\yicarus}[3]{ #1, {Icarus,} {#2}, #3}
\newcommand{\yspd}[3]{ #1, {Sov.\ Phys.\ Dokl.,} {#2}, #3}
\newcommand{\yjcp}[3]{ #1, {J.\ Comput.\ Phys.,} {#2}, #3}
\newcommand{\yjour}[4]{ #1, {#2}, {#3}, #4}
\newcommand{\yjourS}[3]{ #1, {#2}, #3}
\newcommand{\yprep}[2]{ #1, {\sf #2}}
\newcommand{\ybook}[3]{ #1, {#2} (#3)}
\newcommand{\yproc}[5]{ #1, in {#3}, ed.\ #4 (#5), #2}
\newcommand{\pproc}[4]{ #1, in {#2}, ed.\ #3 (#4), (in press)}
\newcommand{\pprocc}[5]{ #1, in {#2}, ed.\ #3 (#4, #5)}
\newcommand{\pmn}[1]{ #1, {MNRAS}, to be published}
\newcommand{\pana}[2]{ #1, {A\&A}, in press, arXiv:#2}
\newcommand{\papj}[1]{ #1, {ApJ}, to be published}
\newcommand{\pgafd}[1]{ #1, {GAFD}, to be published}
\newcommand{\ppapj}[3]{ #1, {ApJ}, {#2}, to be published in the #3 issue}
\newcommand{\sprl}[1]{ #1, {PRL}, submitted}
\newcommand{\sapjl}[3]{:~#1, ``#2,'' {\em Astrophys.\ J.}, submitted, arXiv:#3}
\newcommand{\ssapj}[2]{ #1, {ApJ} (#2)}
\newcommand{\san}[1]{ #1, {Astron.\ Nachr.}, submitted}
\newcommand{\sana}[3]{:~#1, ``#2,'' {\em Astron.\ Astrophys.}, submitted, arXiv:#3}
\newcommand{\ssph}[1]{ #1, {Sol.\ Phys.}, submitted}
\newcommand{\sssph}[2]{ #1, {Sol.\ Phys.} (#2)}
\newcommand{\smn}[1]{ #1, {MNRAS}, submitted}
\hyphenation{ma-ni-fold}
\hyphenation{he-li-ci-ty}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



\def\red{\textcolor{red}}

%Preprint: NORDITA-2018-18

\begin{document}

\titlerunning{Magnetic bipoles in rotating turbulence with coronal envelope}
\authorrunning{Losada et al.}

\title{Magnetic bipoles in rotating turbulence with coronal envelope}

\author{I. R. Losada \inst{1,2,3} \and J. Warnecke\inst{4,1}
\and A. Brandenburg \inst{1,2,5,6}
\and N. Kleeorin \inst{7,1}
\and I. Rogachevskii \inst{7,1}}
\institute{
Nordita, KTH Royal Institute of Technology and Stockholm University,
SE-10691 Stockholm, Sweden\\
\email{illa.rivero.losada@gmail.com}\label{inst1}
\and Department of Astronomy, AlbaNova University Center,
Stockholm University, SE-10691 Stockholm, Sweden \label{inst2}
\and Nordic Optical Telescope, La Palma, Canary Islands, Spain
\and Max-Planck-Institut f\"ur Sonnensystemforschung,
Justus-von-Liebig-Weg 3, D-37077 G\"ottingen, Germany \label{inst3}
\and JILA and Department of Astrophysical and Planetary Sciences,
Box 440, University of Colorado, Boulder, CO 80303, USA \label{inst4}
\and Laboratory for Atmospheric and Space Physics,
Box 590, University of Colorado, Boulder, CO 80303, USA \label{inst5}
\and Department of Mechanical
Engineering, Ben-Gurion University of the Negev, POB 653,
Beer-Sheva 84105, Israel \label{inst6}
}
\date{\today,~ $ $Revision: 1.417 $ $}
\abstract{
The formation mechanism of sunspots and starspots is not fully understood.
It is a major open problem in astrophysics.
}{%
Magnetic flux concentrations can be produced by the negative effective
magnetic pressure instability (NEMPI).
This instability is strongly suppressed by rotation.
However, the presence of an outer coronal envelope was previously
found to strengthen the flux concentrations and make them more prominent.
It also allows for the formation of bipolar regions (BRs).
It is important to understand
whether the presence of an outer coronal envelope also changes
the excitation conditions and the rotational dependence of NEMPI.
}{%
We use direct numerical simulations and mean-field simulations.
We adopt a simple two-layer model of turbulence that mimics the jump
between the convective turbulent and coronal layers below and above the
surface of a star, respectively.
The computational domain is Cartesian and located at a certain latitude
of a rotating sphere.
We investigate the effects of rotation on NEMPI by changing the Coriolis number,
the latitude, the strengths of the imposed magnetic field, and the box
resolution.
}{%
Rotation has a strong impact on the process of BR formation.
Even rather slow rotation is found to suppress their formation.
However, increasing the imposed magnetic field strength also makes the
structures stronger and alleviates the rotational suppression somewhat.
The presence of a coronal layer itself does not significantly reduce
the effects of rotational suppression.
}{}
\keywords{Magnetohydrodynamics (MHD) -- turbulence -- dynamo -- Sun:
  magnetic fields -- Sun: rotation-- Sun: activity
}

\maketitle

\section{Introduction}

The solar dynamo operates in hydromagnetic turbulence in the presence of strong
stratification---especially near the surface.
The stratification can lead to a secondary instability, in
addition to the primary dynamo instability, and can concentrate the field
further into spots.
This instability was discovered by \cite{Kleeorin89,KRR1990}
and applied to explain sunspot formation
and other hydromagnetic processes
in the Sun \citep{KMR93,KMR96,KR94,RK07}.
In the last 10 years, direct numerical simulations (DNS) have demonstrated that this
instability---the negative effective magnetic pressure instability (NEMPI)---is
able to concentrate magnetic fields in different physical environments.

Since its first detection in a stably stratified isothermal setup with
a weak horizontal \citep{BKKMR11} and a weak vertical \citep{BKR2013}
magnetic field, NEMPI demonstrated its operation in polytropic
stratification \citep{LBKR2014} and turbulent convection
\citep{KaBKMR12,KBKKR16}, in the presence of weak
rotation \citep{LBKMR2012,LBKR2013a}, and with a coronal envelope
\citep{WLBKR2013,WLBKR2015}.
In this context, a ``weak'' magnetic field is one that is
of subequipartition strength, but already dynamically important.
Likewise, weak rotation means that the angular velocity is small compared
with the inverse correlation time of the turbulence,
although it is already dynamically important.
The coronal envelope is crucial for producing a bipolar region (BR)
in the domain.
Here we define BRs as structures that are much larger than the
size of individual turbulent eddies.
The emergence of opposite polarities is a consequence
of zero total vertical flux across a horizontal surface.
This implies that regions with weak
large-scale magnetic field are separated by
regions with strong fields
of opposite magnetic polarity.

In principle, spot structures can also be tripolar or quadrupolar
(for example AR~11158), but those cases are rare. Interestingly,
these regions allow one to the determine the magnetic helicity in the
space above by only using the surface magnetic field \citep{BB18}.
This property may help tying the nature of dynamo-generated subsurface
magnetic fields to observations.
A similarly useful tool is helioseismology, which may allow for a detection
of subsurface magnetic fields in the days prior to active regions formation;
see \cite{SRB16} for such work providing evidence for a gradual buildup
of active regions rather than a sudden buoyant emergence from deep down.

The coronal envelope allows the orientation of a weak imposed horizontal
magnetic field to change locally due to the interface between the turbulence zone
and the coronal envelope so that it can attain a vertical component.
On the other hand, rotation plays against the instability (i.e., NEMPI),
which cannot survive above a certain critical rotation rate.
This critical rate was found to be surprisingly
small \citep{LBKMR2012,LBKR2013a}.

It should be mentioned that other types of surface magnetic flux
concentrations have been seen in a number of different circumstances,
all of which share the presence of a strong density stratification.
There is first of all the phenomenon of magnetic flux segregation into
weakly convecting magnetic islands within nearly field-free convecting
regions \citep{TaoWeiss98}.
This has also been seen in several recent high resolution and high
aspect ratio simulations \citep{KapyBeijing12,KBKKR16} and perhaps also
in those of \cite{MS16}.
This process may explain the formation of flux concentrations seen in
the simulation of \cite{SN2012}, in which an unstructured magnetic field
of $1\kG$ is allowed to enter the computational domain at the bottom.
These simulations include realistic surface physics in a domain $96\Mm$
wide and $20\Mm$ deep, so, again, the aspect ratio is large
and there is significant scale separation.
Another approach is the let a flux tube rise from the bottom of the computational
domain to simulate flux emergence at the surface
\citep{FFD93,Fan01,Arc04,Arc05,FAZS17};
see \cite{Fan09} for a review.
Similar results, but with realistic surface physics,
have been obtained by \cite{CRTS10} and \cite{rempel2014}, who let a
semi-torus of magnetic field advect through the lower boundary.
Their simulations showed that the magnetic field is able to rise through
the top $16\Mm$ of the convection zone to form spots; see the reviews by
\cite{CI14} and \cite{SAP14}.
In their simulations, however, flux emergence is significantly faster
than in the real Sun \citep{BSBCGLR16}.
Recently, \cite{CRF17} were able to reproduce a complex sunspot
emergence using a modified magnetic flux bundle from the dynamo
simulation of \cite{Fan2014} in a spherical shell
and inserting it into a setup similar to that of \cite{rempel2014}.

The process of magnetic flux concentration in convection could be related
to the magnetic suppression of the convective heat flux, which, again,
could lead to a large-scale instability \citep{KitchatinovMazur2000}.
On the other hand, \cite{KKWM2010} explain the formation of magnetic flux
concentrations in their radiation-hydromagnetic simulations as being
confined by the random vorticity associated with convective downdrafts.
However, the process seen in their simulations seems to be similar to
flux concentrations found in the downdraft of the axisymmetric simulations
of \cite{GW81}.
There is also a process known as convective collapse \citep{Parker78},
which can lead to a temporary concentration of field from a weaker, less
concentrated state to a more concentrated collapsed state, e.g., from
$1270\G$ to $1650\G$ in the specific calculations of \cite{Spruit79}.
However, the collapsed state is not in thermal equilibrium, so the
system will slowly return to an uncollapsed state.
This effect may be related to the ionization physics, which can strongly
enhance the resulting concentrations \citep{BhatBrandenburg15}.

Strong magnetic flux concentrations have also been seen in simulations
where a large-scale dynamo is responsible for generating magnetic field.
The dynamo arises as a result of helically driven turbulence in the lower
part of the domain, while in the upper part turbulence is nonhelically driven.
In the upper part, the magnetic field displays the formation of strongly
concentrated BRs.
This was seen in DNS in both Cartesian domains
\citep{MBKR14,Jabbari16,Jabbari2017} and spherical shells
\citep{Jabbari15}.

The relation to NEMPI is unclear in some of these cases, because NEMPI
can be excited when the magnetic field is below the equipartition value
of the turbulence.
A negative effective magnetic pressure is possible in somewhat
deeper layers and at intermediate times, and that may be important
in initializing the formation of magnetic flux concentrations.
In particular, it would lead to downward suction along vertical
magnetic field lines which creates an underpressure in the upper parts
and results in an inflow.
The latter causes further concentrations in the upper parts.
This was clearly seen in the axisymmetric mean-field simulations (MFS)
of \cite{BGJKR14}; see \cite{BRK2016} and \cite{LWGRBKR17} for recent
reviews.

In the present work, we consider a setup similar to that of
\cite{WLBKR2013,WLBKR2015}.
There, turbulence of an isothermal gas is
forced in the lower part of a horizontally periodic domain, while the
upper part is left unforced and subject to the response from the
dynamics of the lower part.
This approach has been used to study the effect of a coronal
envelope on the dynamo \citep[e.g.][]{WB2010,WBM2011,WKMB13,WKKB16}
and the formation of coronal ejections \citep{WBM12,WKMB12}.
In these simulations, the simple treatment of the coronal envelope does
not allow for a low plasma $\beta_{\rm c}$ as in the solar corona,
where $\beta_{\rm c}$ is the ratio of magnetic pressure to gas
pressure. However, also in the solar corona the value of
$\beta_{\rm c}$ is not extremely small \citep[e.g.][]{PWCC15} and plasma flows
can play an important role for the formation of loop structures \citep{WCBP17}.

We include the Coriolis force to examine the effects of rotation in
the presence of our simplified corona to study whether this facilitates
the development and detectability of NEMPI, and whether it changes the
critical growth rate above which NEMPI is suppressed.
We also study the dependence on latitude, as well as the dependence on
the numerical resolution.
Finally, we compare our solutions with corresponding MFS, in which
a prescribed effective (mean-field) magnetic pressure operates only
beneath the surface, but not in the coronal layer.

\section{The model}

\subsection{DNS}
\label{Sec:DNS}

We use the same two-layer model as \cite{WLBKR2013,WLBKR2015}.
We considered a Cartesian domain with forced turbulence in the
lower part (referred to as turbulent layer), and a more quiescent
upper part (referred to as coronal envelope).
We further adopt an isothermal equation of state and
solve the equations for the velocity $\UU$,
the magnetic vector potential $\AAA$, and the density $\rho$.
We adopt units for the magnetic field such that
the vacuum permeability is unity.
Here we extend this model by including the presence of rotation with an
angular velocity $\Omega$,
\begin{equation}
{\DD\UU\over\DD t}=-2\OO\times\UU
-\cs^2\nab\ln\rho+\grav+{1\over\rho}\JJ\times\BB+\Theta_w(z)\ff+\FF_\nu,
\label{DUDt}
\end{equation}
\begin{equation}
{\partial\AAA\over\partial t}=\UU\times\BB-\eta\JJ,
\end{equation}
\begin{equation}
{\partial\rho\over\partial t}=-\nab\cdot(\rho\UU),
\end{equation}
where $\DD/\DD t=\partial/\partial t+\UU\cdot\nab$ is the advective
derivative, $\BB=\BB_0+\nab\times\AAA$ is the magnetic field,
$\BB_0=(0,B_0,0)$ is a weak imposed uniform field in the $y$ direction,
$\JJ=\nab\times\BB$ is the current density,
$\FF_\nu=\nab\cdot(2\nu\rho\SSSS)$ is the viscous force,
$\nu$ is the kinematic viscosity,
$\eta$ is the magnetic diffusivity,
$\grav$ is the gravitational acceleration,
${\sf S}_{ij}=\half(\partial_j U_i+\partial_i U_j)-\onethird\delta_{ij}\nab\cdot\UU$
is the traceless rate-of-strain tensor, and $\ff$ is a forcing function
that consists of random, white-in-time, plane, nonpolarized waves within
a certain narrow interval around an average wavenumber $\kf$.
It is modulated by a profile function $\Theta_w(z)$,
\begin{equation}
\label{eq:error}
\Theta_w(z)=\half\left(1-\erf{z\over w}\right),
\end{equation}
that ensures a smooth transition between unity in the lower layer and
zero in the upper layer.
Here $w$ is the width of the transition.
The angular velocity vector $\OO$ is quantified by its modulus
$\Omega$ and colatitude $\theta$, such that
\EQ
\OO=\Omega\left(-\sin\theta, 0, \cos\theta\right).
\EN

Following \cite{WLBKR2013}, the domain used in the DNS is
$L_h\times L_h\times L_z$, where $L_h=2\pi$ and $L_z=3\pi$ with
$-\pi\leq z\leq 2\pi$.
This defines the base horizontal wavenumber $k_1=2\pi/L_h$,
which is set to unity in our model.
Our Cartesian coordinate system $(x,y,z)$ corresponds to a local
representation of a point on a sphere mapped to spherical
coordinates $(r,\theta,\phi)\to(z,x,y)$, where $r$ is radius,
$\theta$ is colatitude, and $\phi$ is longitude.
Similar to earlier work \citep{Kemel12b,Kemel13a}, we use
in all cases $\kf=30\,k_1$ and $\nu=10^{-4}\,\cs/k_1$ with the sound
speed $\cs$.
The normalized gravity is given by $g H_\rho/\csq=1$, which is just
slightly below the value of $1.2$ that was found to maximize the
amplification of magnetic field concentrations \citep{WLBKR2015}.
Here $H_\rho$ is the density scale height.
As in most of our earlier work, we use $k_1 H_\rho=1$, so the
vertical density contrast is $\exp(L_z/H_\rho)=\exp3\pi\approx12,000$.
For the width of the profile functions in the DNS and MFS,
we use $k_1 w=0.05$.

For the rms velocity, we will use the averaged value in the turbulent
layer defined as: $\urms=\bra{\UU^2}_{xy;z\le 0}^{1/2}$.
We normalize the magnetic field by its equipartition value,
$\Beq=\sqrt{\rho} \, \urms$, using either the $z$ dependent value of
$\Beq$ or the value at the surface at $z=0$, i.e., $\Beqz\equiv\Beq(0)$.

Our simulations are characterized by the magnetic and fluid Reynolds numbers,
\EQ
\Rm=\urms/\eta\kf,\quad
\Rey=\urms/\nu\kf,
\EN
respectively, the Coriolis number
\EQ
\Co=2\Omega\tau,
\label{CoriolisNum}
\EN
where $\tau=1/\urms\kf$ is the eddy turnover time,
and the colatitude $\theta$ of our domain is positioned on the sphere.
In the following, we use $\Rm\approx14$ and $\Rey\approx29$,
so the magnetic Prandtl number is $\Pm=\nu/\eta=0.5$.
These values of the magnetic and fluid Reynolds numbers
are based on the forcing wavenumber, which is rather high
($\kf/k_1=30$).
Thus, the values of $\Rm$ and $\Rey$ based on the wavenumber of the
domain would be 420 and 870, respectively, and those based on the size
of the domain, which are larger by another factor of $2\pi$, would be
2640 and 5470, respectively.
The definitions of these Reynolds numbers must therefore be kept
in mind when comparing with other work.
In our definition, the magnetic Reynolds number required for the
effective magnetic pressure to be negative must be larger than a
critical value of about three \citep{BKKMR11,BKKR12,KaBKMR12,WLBKR2015}.
For small $\Rm$, the effective magnetic pressure can only be positive
\citep{RKS12,BKKR12}.
In our work, we choose $\Rm$ to be about ten times supercritical.
With the choice of $\Rey\approx29$ and $\Pm=0.5$, we exclude that the excitation of a
small-scale dynamo influences our results. As shown in
%\cite{WLBKR2015}, $\Pm=1$ is needed to excite a small-scale dynamo
\cite{WLBKR2015}, $\Pm\ge1$ is needed to excite a small-scale dynamo
in this setup.
%JW: This is more accurate.

In this work, time is often expressed in units of the turbulent diffusive
time, $\tautd=\etatz k_1^2$, where $\etatz=\urms/3\kf$ is an estimate
of the turbulent magnetic diffusivity.
As in earlier work \citep{WLBKR2013,WLBKR2015}, we use the Fourier
filtered magnetic and velocity fields as diagnostics for characterizing
large-scale properties of the solutions.
Our Fourier filtered fields are denoted by an overbar and the superscript
$\rm fil$, i.e., $\meanf{B_z}$ for the vertical magnetic field.
This includes contributions with horizontal wavenumbers below $\kf/2$.
This filtering wavenumber is the same as that used in \cite{WLBKR2013},
but the cutoff wavenumber is three times larger than the filter value
$\kf/6$ used by \cite{BKR2013} and \cite{WLBKR2015}.
This is appropriate here because, owing to the nature of our BRs,
where the spots tend to be close together, they would not be
well captured when the averaging scale is too large or the filtering
wavenumber too small.
The corresponding spectral magnetic energy contained in the
vertical magnetic field, $B_z$, is $E_{\rm M}^z$ and obeys
$\int E_{\rm M}^z\,\dd k=\bra{B_z^2}/2$.
Of particular interest is the energy per logarithmic wavenumber
interval, $2k_\ast E_{\rm M}^z(k_\ast)$, which we usually evaluate at
$k_\ast/k_1=2$, where the energy reaches a maximum.
(The factor $2$ in front of $kE_{\rm M}^z$ compensates for the 1/2 factor
in the definition of the energy.)
For the velocity, however, we find that $\kf/6$ is the appropriate
filtering wavenumber.
Therefore, we filter the velocities on a larger scale than the magnetic
field.

We compute growth rates and magnetic energies as averages over a certain
time interval.
We compute error bars as the largest departure from any one third
of the full time interval used for computing the average.
In some cases, those error estimates were themselves unreliable.
In such exceptional cases we have replaced it by the average error
for other similar simulations.

We use resolutions between $192\times192\times384$ and
$1152\times1152\times2304$ meshpoints in the $x$, $y$,
and $z$ directions, respectively.
We adopt periodic boundary conditions in the $xy$ plane, a stress-free
perfect conductor condition on the bottom boundary, and a stress-free
vertical field condition on the top boundary.

\subsection{MFS}

In this section, we state the relevant equations for the mean-field
description of NEMPI in a system with coronal envelope
in the presence of rotation.
The relevant equations have been obtained by
\cite{Kleeorin89,KRR1990,KMR96} and \cite{KR94}
through ensemble averaging.
In practical applications, these averages should be replaced by spatial
averages, but their precise nature depends on the physical circumstances
and could be planar (e.g., horizontal) or azimuthal (e.g., around a
flux tube), or some kind of spatial smoothing.
In the present case with inclined stratification, rotation vectors, and
mean magnetic field vectors, we expect the formation of bipolar structures
that cannot be described by simple planar or azimuthal averages.
The only meaningful average is a smoothing operation that could preserve
such structures.
In the following, we denote the dependent variables of our MFS by overbars.
For the analysis of our DNS, on the other hand, we use sometimes Fourier
filtering and sometimes horizontal averaging and denote them also by
an overbar.
In those cases, a corresponding comment will be made, and in the particular
case of Fourier filtering, the variable will be additionally denoted by the
superscript ``fil''.

In the MFS, the equations for the mean velocity $\meanUU$, mean vector
potential $\meanAA$, and mean density $\meanrho$, are given by
\begin{equation}
{\meanDD\,\meanUU\over\DD\, t}=-2\OO\times\meanUU
-\cs^2\nab\ln\meanrho+\grav+\meanFFFF_{\rm M}+\meanFFFF_{\rm K},
\label{dUmean}
\end{equation}
\begin{equation}
{\partial\meanAA\over\partial t}=\meanUU\times\meanBB-(\etat+\eta)\meanJJ,
\end{equation}
\begin{equation}
{\partial\meanrho\over\partial t}=-\nab\cdot(\meanrho\meanUU),
\end{equation}
where $\meanDD/\DD t=\partial/\partial t+\meanUU\cdot\nab$
is the advective derivative based on $\meanUU$,
$\etat$ is turbulent magnetic diffusivity,
\EQ
\meanFFFF_{\rm K}=(\nut+\nu)\left(\nabla^2\meanUU+\onethird\nab\nab\cdot\meanUU
+2\meanSSSS\nab\ln\meanrho\right)
\EN
is the total (turbulent plus microscopic) viscous force
with $\nut$ being the turbulent viscosity,
$\meanSSS_{ij}=\half(\meanU_{i,j}+\meanU_{j,i})
-\onethird\delta_{ij}\nab\cdot\meanUU$
is the traceless rate-of-strain tensor of the mean flow
and, as in the DNS, we adopt units for the mean magnetic
field such that the vacuum permeability is unity.
The effective Lorentz force, $\meanFFFF_{\rm M}$,
which takes into account the turbulence contributions,
i.e., the effective magnetic pressure \citep{KMR96,RK07,BRK2016}
and an anisotropic contribution resulting from gravitational
stratification, is given by
\EQ
\meanrho \, \meanFFFF_{\rm M} = \meanJJ\times\meanBB
+\nab\left[\half q_{\rm p}(z,\meanBB)\,\meanBB^2\right]
+\zzz{\partial\over\partial z}\left[q_{\rm g}(z,\meanBB)\,\meanBB^2\right],
\label{efforce}
\EN
where $q_{\rm p}$ and $q_{\rm g}$ are functions that have previously
been determined from DNS \citep{BKKR12}.
\cite{WLBKR2015} found $q_{\rm g}$ to be negative for weak and moderate
stratification (note that the abscissa of their Fig.~6 shows $g/k_1\cs^2$
and not $gH_\rho/\cs^2$, as was incorrectly written).
Thus, there is the possibility of partial cancelation, which we model
here by assuming $q_{\rm p}$ and $q_{\rm g}$ to have the same profile with
\EQ
q_{\rm g}=a_{\rm g} q_{\rm p}\quad\mbox{($a_{\rm g}=\const$)}.
\label{adef}
\EN
\cite{WLBKR2015} determined $q_{\rm p}\beta^2=-q_{\rm
  g}\beta^2=-0.002$, resulting in $a_{\rm g}=-1$ for the same
stratification as in this work ($g/k_1\cs^2=1$).
We note here that the values of $q_{\rm p}$ and $q_{\rm g}$ are
the result of averaging in time and space,
so locally the values can be different, and therefore they do
not need to cancel out locally.
Furthermore, the error estimate by the spread is comparable to the
averaged value, see Fig.~6 of \cite{WLBKR2015}.
We model $q_{\rm p}$ and $q_{\rm g}$ as the product of a part that
depends only on $\beta^2=\meanBB^2/\Beq^2(z)$ and a profile function
$\Theta_w(z)$.
The latter function
varies only along the $z$ direction and it mimics the effects of the coronal
layer, using the same error function as in \Eq{eq:error}, i.e.,
\EQ
q_{\rm p}(\meanBB,z) = q_{\rm p}^{(B)}(\beta^2)\,\Theta_w(z),
\label{parameterization}
\EN
where
\EQ
q_{\rm p}^{(B)}(\beta^2)= {\qpz\over1+\beta^2/\betap^2}
={\beta_\ast^2\over\betap^2+\beta^2}.
\label{param_beta}
\EN
Here $\beta_\ast=\sqrt{\qpz}\,\betap$ is a parameter that can be used
alternatively to $\qpz$ and has the advantage that the growth rate of
NEMPI is predicted to be proportional to it \citep{Kemel13a}.
In the following, we mainly use the parameters found by \cite{LBKR2013a},
namely $\qpz=32$ and $\betap=0.058$, which corresponds to
$\beta_\ast=0.33$.
On one occasion, we also use another parameter combination that will be
motivated by our results presented in \Sec{EffectiveMagneticPressure}
below.
In hindsight, it might have been more physical to use $\Theta_w^2$ in
\Eq{parameterization}, but we know from experience that this would hardly
make a noticeable difference.

We note in passing that, while in both the DNS and the MFS, the coronal
envelope is modeled with the same profile function $\Theta_w(z)$,
in the MFS, it appears underneath the gradient
(inside the effective Lorentz force), while in the DNS it does not.
In the MFS, this leads to an additional term involving the derivative
of $\Theta_w(z)$, which is a gaussian function.
In one of the cases reported below, we checked that the presence of this
term causes a very minor difference in the growth rates.

In addition to adopting the parameterization given by \Eq{parameterization},
the effects of turbulence are modeled in terms of turbulent viscosity $\nut$
and turbulent magnetic diffusivity $\etat$.
Both can be expressed in terms of $\urms$ and $\kf$, whose values are
known from the DNS and give $\nut=\etat\approx10^{-3}\cs H_\rho$.
Thus, $3\etat/\eta=\Rm$ and $3\nut/\nu=\Rey$.
Among the range of other possible mean-field effects, we have included
here only the negative effective magnetic pressure
functions $q_{\rm p}$ and $q_{\rm g}$.

As already noted by \cite{LBKMR2012}, the usual Coriolis number
is not the relevant quantity characterizing the relative importance
of rotation on NEMPI.
A more meaningful quantity is the ratio of $2\Omega/\lambda_{\ast0}$,
where $\lambda_{\ast0}$ is the nominal
value of the growth rate
\EQ
\lambda_{\ast0}=\beta_\ast\urms/H_\rho,
\label{lamast0}
\EN
which was found to be comparable to the growth rate of NEMPI
\citep{LBKMR2012}.
Therefore, the Coriolis number given by \Eq{CoriolisNum},
can be rewritten as
\EQ
\Co={2\Omega\over\urms\kf}
={2\Omega\over\lambda_{\ast0}}\,{\beta_{\ast}\over \kf H_\rho}.
\label{Omlam}
\EN
With $\urms\approx0.1\cs$ and $k_1 H_\rho=1$, this means that
$\lambda_{\ast0}=0.033\cs k_1$; see \Eq{lamast0}.
This value will also be used to characterize the rotation rate
of the DNS, which are well characterized by the parameter
$\beta_\ast=0.33$; see \cite{BKKR12}.
Thus, $2\Omega/\lambda_{\ast0}$ is about 100 times larger than $\Co$.
Note also that in the MFS, $\Co=6\Omega\etat/\urms^2$
\citep{Jabbari2014}, which results in the same estimate.
The actual growth rates obtained from our DNS and MFS will be
normalized either also by $\lambda_{\ast0}$ or by $\tautd^{-1}$.

Both DNS and MFS simulations are done with the
\PCn\footnote{\tt{http://github.com/pencil-code}}.
It uses sixth order accurate finite differences and a third-order
timestepping scheme.
It comes with a special mean-field module that can be invoked for the
MFS.

\begin{table*}[t!]\caption{
Summary of run with different resolutions.
}\vspace{10pt}
\centerline{
\begin{tabular}{lccccccccccccc}
Run &  Resolution  & $B_0/\Beqz$
& $\Bfm/\Beqz$
&$\tautd^{\meanB_z=\max}$
&$(E_{\rm M}^{z,\max}/\Beqz^2)^{1/2}$
&$\tautd^{E_{\rm M}^z=\max}$
& $\Rey_{grid}$ &$\lambda\tautd$ &BR\\
\hline
\hline
A1 & $192^2\times384$  & 0.027 & 0.37 &1.2&0.06 (0.08) & 0.61 (1.6) & 0.018 &$0.74 \pm 0.08$ & yes \\%
A2 & $384^2\times768$  & 0.026 & 0.35 &0.9&0.08 (0.11) & 0.92 (0.9) & 0.018 &$1.28 \pm 0.13$ & yes \\%
A3 & $576^2\times1152$ & 0.025 & 0.42 &1.2&0.07 (0.10) & 0.61 (1.2) & 0.074 &$1.27 \pm 0.11$ & yes \\%
A4 & $768^2\times1536$ & 0.025 & 0.54 &1.5&0.13 (0.17) & 1.41 (1.4) & 0.001 &$1.27 \pm 0.14$ & yes \\%
A5 & $1152^2\times2304$& 0.025 & 0.40 &1.0&0.09 (0.12) & 0.92 (0.9) & 0.004 &$1.26 \pm 0.10$ & yes \\%
\hline
\hline
\label{runs1}
\end{tabular}}
\tablefoot{$\Bfm$ is the maximum of the Fourier-filtered vertical
  magnetic field strength at the surface ($z=0$),
  $\tautd^{\meanB_z=\max}$ is the time when $\Bfm/\Beqz$ reaches a
  maximum while $\tautd^{E_{\rm M}^z=\max}$ is the time when
$E_{\rm M}^z(k_\ast)$ with $k_\ast/k_1=2$ reaches a maximum.
The numbers in parentheses refer to spectral values averaged between
$k/k_1=1$ and $4$.
$\Rey_{\rm grid}= \urms / \nu k_{\rm Ny}$ is the mesh Reynolds number,
  $\Peffmin$ is the minimum value of the effective magnetic pressure,
  defined in \Sec{EffectiveMagneticPressure}.
The column BR indicates the visual appearance of BRs at the surface.
}
\end{table*}

\section{DNS results}

\subsection{Numerical resolution}
\label{Resol}

Since NEMPI is a mean-field instability, which relies on small-scale
turbulence for developing large-scale structures, we begin by
demonstrating the effects of changing the resolution on the formation
of magnetic field concentrations.
The growth rates are shown in \Fig{growth_vs_res} as a function
of resolution.
The corresponding simulations are listed in \Tab{runs1} where we have
always used the same domain size of $(2\pi)^2\times3\pi$, but with
different numbers of meshpoints.
We also quote the mesh Reynolds number,
$\Rey_{\rm grid}= \urms / \nu k_{\rm Ny}$, where
$k_{\rm Ny}=\pi/\delta x$ is the Nyquist wavenumber
and $\delta x$ is the mesh spacing.
In all cases, this number is well below unity.
Sometimes the quantity $\urms \delta x/ \nu$ is used in the
literature; it is simply $2\pi$ times larger than our $\Rey_{\rm grid}$.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pLambda_reso}
\end{center}\caption[]{
Growth rate $\lambda$ (black line), the peak spectral
magnetic field $(2k_\ast E_{\rm M}^{z,\max})^{1/2}$ (red), and the maximum of
the Fourier-filtered vertical magnetic field $\Bfm$ (blue) for Runs~A1-A5 at
different resolutions.
}\label{growth_vs_res}\end{figure}

\begin{figure}[t!]\begin{center}
\includegraphics[width=0.9\columnwidth]{bb3_xy_resolution}
\end{center}\caption[]{
Vertical magnetic field $B_z$ at the surface for runs with
different resolution (Run~A1 to Run~A3 and A5)
at the time when the structures are strongest.
}\label{fig:xy_resolution}\end{figure}

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{meanb_vs_t_growth_384k30_B002_Om0_beskow_gnu}
\end{center}\caption[]{
Evolution of $\meanB_z^{\max}/\Beq$ vs.\ $t/\tautd$ for Run~A2
showing exponential growth with growth rate $\lambda=1.3\tautd^{-1}$
(black solid line), compared with that of $B_z^{\rm rms}/\Beq$
(red dashed line) and $B_z^{\max}/\Beq$ (blue solid line) at
$z=0$.
We also show the unity line corresponding to $\Beq$ (orange dotted line).
}\label{meanb_vs_t_growth}\end{figure}

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{prslice_single}
\includegraphics[width=\columnwidth]{prslice_single_1152k30_B002_Om0_beskow_gnu2}
\end{center}\caption[]{
Vectors of the mean (Fourier-filtered) flow superimposed
on a color scale representation of $B_z$ at the surface at
$t/\tautd=0.91$  for Run~A4 (upper panel) and
$t/\tautd=1.12$ for Run~A5 (lower panel).
Note that the coordinate system has been rotated by $90\degr$,
so $y$ points to the right and $X$ points downward.
}\label{prslice_single}\end{figure}

We see formation of bipolar regions (denoted by BR in the table) in all
the cases, but at the lowest resolution,
the growth rate of the magnetic field is significantly
smaller than at all higher resolutions.
Based on this, we conclude that a resolution of $384^2\times768$ meshpoints
results in a good compromise between accuracy and computational cost.
Therefore, we use simulations with this resolution for the following parameter study.
Because of this reason, \cite{WLBKR2015} doubled their resolution in
their followup work in \cite{WLBKR2013}.
Qualitatively, the coherence increases with increasing
resolution; see \Fig{fig:xy_resolution}.
We see from \Tab{runs1} the structures become strongest at a resolution
of $768^2\times1536$ meshpoints, where the normalized Fourier-filtered vertical
magnetic field at the surface, $\Bfm/\Beqz$, reaches
a value of 0.54 corresponding to only 0.4 at both lower and higher
resolutions.

\subsection{Growth of BRs}

We now discuss the main properties of BRs.
Large-scale BRs form during the first one or two
turbulent-diffusive times.
They are referred to as large-scale structures, because their size
is that of many turbulent eddies (about $2\pi/\kf$ in the horizontal
plane); see \Fig{prslice_single}.
To average over these turbulent eddies and still resolve the
large-scale structure of the BRs, we Fourier-filter the magnetic field
at the surface. To investigate the growth of these structures we then
plot the maximum of the vertical Fourier-filtered magnetic field
$\meanf{B}_z$ over time.
In \Fig{meanb_vs_t_growth}, which corresponds to
Run~A2, we clearly see an exponential growth with growth rate
$\lambda\approx1.3\,\tautd^{-1}$.
A similar growth has been seen before in numerical experiments both
with a horizontally imposed field \citep{BKKMR11, Kemel12b} and a vertical one
\citep{BKR2013}.
A similar value has also been determined with the same setup, where
BRs form \citep{WLBKR2015}.
Such exponential growth is suggestive of NEMPI.

In \Fig{meanb_vs_t_growth}, we also show for comparison the evolution
of the maximum of the surface vertical magnetic field $B^{\rm max}_z$,
i.e., not the filtered value.
Its value is always close to the local equipartition field strength.
No exponential growth phase can be seen in the temporal evolution
of $B^{\rm max}_z$.
Likewise, the rms value of surface vertical magnetic field
$B_z^{\rm rms} =\bra{B^2_z(z=0)}_{xy}^{1/2}$ is about 20 times smaller
and shows no exponential growth.

The formation of BRs in our simulations are associated with large-scale
flows.
As for the magnetic field, we perform Fourier filtering to
averaged over the turbulent scales.
In \Fig{prslice_single} we show these Fourier-filtered large-scale
flows that includes only wavenumbers below $\kf/6$.
The two panels are for Runs A4 and A5
with two different resolutions, but otherwise the same as
Run~A2, and similar times.
In both cases there are BRs, but in one case the two spots
are more separated from each other.
Nevertheless, in all cases the BRs are surrounded by a
large-scale inflow with relative rms value
$\meanU_{\rm hor}^{\rm rms}/\urms=0.16$--$0.18$,
where $\meanU_{\rm hor}^{\rm rms}=\bra{U_x^2+U_y^2}_{xy}^{1/2}$.
Its maximum value is around $0.4$.
Given that the scale of the turbulent motions
is much smaller than the scale of the spots, and that there is otherwise no
mechanism producing large-scale flow perturbations,
the inflow can only
be a consequence of the magnetic field itself, and not the other way around.
This is indeed also what one expects for NEMPI.

Another diagnostics for the formation of magnetic structures in
our simulations are magnetic power spectra taken at the surface.
In \Fig{pspec} we can see the time evolution of such power spectra.
At early times when there are no structures yet, the spectrum peaks at
the energy injection wavenumber,
$\kf=30\,k_1$.
As time evolves and structures start forming near the surface,
magnetic energy is transported toward smaller wavenumbers.
When we see the BRs forming at the surface, the magnetic
power spectrum peaks at $k=k_\ast\approx2k_1$, and
the amplitude at this wavenumber decreases until the structures disappear.
The strength of the magnetic surface structures at this wavenumber is characterized
by the value of $(2kE_{\rm M}^{z,\max})^{1/2}$ at $k=k_\ast=2\,k_1$; see
\Sec{Sec:DNS}.
The resulting values are listed in \Tabss{runs1}{runs3}.
In addition, we also give the spectral values averaged over the first
four wavenumbers from $k/k_1=1$ to $4$.
The averaging helps reducing the sensitivity to discretization noise
that arises from looking at just one wavenumber $k_\ast$.
In the tables, we also judge BR formation as yes, no, or weak.
These attributes are based on the qualitative assessment of images
of $B_z$.

\begin{figure}[t!]
\centering
\includegraphics[width=\columnwidth]{pBz_spec_384k30_B002_Om0}
\label{fig:pspecb}
\caption{Power spectra of $B_z$ at the surface ($z=0$) for Run~A2 at
several
times around the maximum growth for zero rotation.
The inset shows the time dependence (time in units of $\tautd$)
for $k=k_\ast=2\,k_1$ and for the values averaged between $k/k_1=1$ and $4$.
}\label{pspec}
\end{figure}

\begin{table*}[t!]\caption{
Summary of runs with different Coriolis numbers and colatitudes.
}\vspace{10pt}
\centerline{
\begin{tabular}{lccccccccc}
Run &   $\Co$ & $\theta$
   & $\Bfm/\Beqz$
   &$\tautd^{\meanB_z=\max}$
   &$\sqrt{2k_\ast E_{\rm M}^{z,\max}}/\Beqz$
   &$\tautd^{E_{\rm M}^z=\max}$
   &$\lambda\tautd$ &$\Peffmin$&BR\\
\hline
\hline
A2 & 0 & 0 &  0.35 &0.9&0.08 (0.11) & 0.92 (0.9) & $1.28 \pm 0.13$ &$-0.029$& yes \\%
\hline
% 384k30_B002_Om002_theta0
B1 & 0.0012 & 0  & 0.31 & 0.8 &0.17 (0.22) &0.61 (0.61)&$1.32\pm 0.20$ &$-0.029$& yes \\
B2 & 0.0012 & 30 & 0.35 & 0.9 &0.13 (0.18) &0.61 (0.73)&$1.37\pm 0.11$ & & yes \\
B3 & 0.0012 & 60 & 0.32 & 1.7 &0.16 (0.20) &0.67 (0.67)&$1.46\pm 0.16$ & & yes \\
B4 & 0.0012 & 90 & 0.42 & 1.8 &0.13 (0.16) &0.73 (0.67)&$0.58\pm 0.06$ & & yes
\\
\hline
% 384k30_B002_Om004_theta0
C1 &  0.0023 & 0 & 0.672 & 1.7 &0.30 (0.42) &1.59 (1.59)& $1.03\pm 0.07$ &$-0.025$& yes\\
C2 &  0.0023 & 30& 0.303 & 0.8 &0.12 (0.16) &0.61 (0.61)& $1.46\pm 0.19$ &$-0.021$& yes\\
C3 &  0.0023 & 60& 0.428 & 1.7 &0.18 (0.26) &1.65 (1.65)& $0.66\pm 0.06$ &$-0.019$& yes\\
C4 &  0.0023 & 70& 0.341 & 2.4 &0.12 (0.16) &3.18 (1.96)& $0.24\pm 0.05$ & & yes\\
C5 &  0.0023 & 90& 0.467 & 2.0 &0.14 (0.20) &1.16 (1.96)& $0.64\pm 0.06$ &$-0.020$& yes\\
\hline
% 384k30_B002_Om005_theta0
D1 &  0.0029 & 0 & 0.532 & 1.41 &0.26 (0.38) & 1.16 (1.28)& $1.40\pm 0.17$ &$-0.028$& yes\\
D2 &  0.0029 & 60& 0.406 & 2.63 &0.16 (0.22) & 1.16 (1.96)& $0.37\pm 0.05$ & & yes\\
\hline
% 384k30_B002_Om006_theta
E1 & 0.0035 & 0 & 0.228 & 0.6 &0.11 (0.15) & 0.9 (0.9) &$0.92\pm 0.3$ &$-0.023$& yes\\
E2 & 0.0035 & 60& 0.241 & 1.1 &0.13 (0.16) & 0.5 (0.5) &$0.56\pm 0.1$ & & weak \\
\hline
% 384k30_B002_Om01_theta0
F1 &  0.0076 & 0 & 0.296 & 1.4&0.11 (0.14) & 1.5 (1.5)& $0.38\pm 0.09$ &$-0.029$& weak\\
F2 &  0.0076 & 30& 0.269 & 3.0&0.12 (0.16) & 1.8 (1.8)& $0.17\pm 0.05$ & & yes\\
F3 &  0.0076 & 60& 0.226 & 0.5&0.11 (0.14) & 0.3 (0.3)& $0.16\pm 0.13$ & & no\\
F4 &  0.0076 & 90& 0.209 & 0.7&0.11 (0.15) & 0.8 (0.8)& $0.60\pm 0.15$ & & weak
\\
\hline
\hline
\label{runs2}
\end{tabular}}
\tablefoot{
All runs have a resolution of $384^2\times768$ meshpoints (as in Run~A2),
an imposed field of $B_0/\Beqz \approx 0.026$,
and a size of $(2\pi)^2\times3\pi$.
The column BR indicates the visual appearance of BRs at the surface.
All other quantities are defined in \Tab{runs1}.
}\end{table*}

\begin{table*}[t!]\caption{
Summary of runs with different values of imposed field
and Coriolis number.
}\vspace{10pt}
\centerline{
\begin{tabular}{lccccccccc}
Run &   $B_0/\Beqz$& $\Co$
   & $\Bfm/\Beqz$
   &$\tautd^{\meanB_z=\max}$
   &$\sqrt{2k_\ast E_{\rm M}^{z,\max}}/\Beqz$
   &$\tautd^{E_{\rm M}^z=\max}$
   &$\lambda\tautd$ & $\Peffmin$ &BR\\
\hline
\hline
A2 & 0.026 & 0 &  0.35 &0.9&0.08 (0.11) & 0.92 (0.9)      &$1.28\pm 0.13$ &$-0.029$& yes \\%
\hline
% 384k30_B005_Om004_theta0
G1 & 0.065 & 0.002 & 0.38 & 0.4& 0.32 (0.40)& 0.49 (0.55) &$1.21\pm 0.20$ &$-0.046$& yes \\%growth384_B005_Om004_th0.ps
G2 & 0.066 & 0.004 & 0.54 & 0.9& 0.32 (0.46)& 0.73 (0.73) &$1.27\pm 0.12$ &$-0.041$& yes \\%growth384_B005_Om006_th0.ps
G3 & 0.066 & 0.006 & 0.52 & 1.6& 0.38 (0.52)& 1.59 (1.59) &$0.67\pm 0.05$ &$-0.039$& yes \\%growth384_B005_Om01_th0.ps
G4 & 0.067 & 0.012 & 0.50 & 2.5& 0.38 (0.50)& 1.34 (1.34) &$0.50\pm 0.06$ &$-0.048$& yes \\%growth384_B005_Om02_th0.ps
G5 & 0.066 & 0.015 & 0.46 & 1.3& 0.32 (0.42)& 0.98 (1.89) &$0.34\pm 0.07$ & & weak \\%growth384_B005_Om025_th0.ps
G6 & 0.068 & 0.018 & 0.43 & 1.4& 0.28 (0.34)& 0.98 (0.98) &$0.33\pm 0.06$ &$-0.036$& no \\%growth384_B005_Om03_th0.ps
\hline
% 384k30_B010_Om004_theta0
H1 & 0.13  & 0.002 & 0.50& 0.8&0.28 (0.38) &0.5 (0.3)& $0.55\pm 0.09$ &$-0.069$& yes\\ %growth384_B010_Om004_th0.ps
H2 & 0.14  & 0.006 & 0.49& 0.5&0.32 (0.42) &0.5 (0.5)& $0.22\pm 0.13$ &$-0.070$& yes\\ % growth384_B010_Om01_th0.ps
H3 & 0.14  & 0.012 & 0.63& 2.3&0.44 (0.64) &0.7 (2.1)& $0.20\pm 0.03$ &$-0.067$& yes\\ %growth384_B010_Om02_th0.ps
H4 & 0.14  & 0.018 & 0.54& 3.0&0.32 (0.54) &3.5 (0.9)& $0.09\pm 0.04$ &$-0.068$& weak\\%growth384_B010_Om03_th0.ps
\hline
% 384k30_B020_Om005_theta0
I1 & 0.27 & 0.002 & 0.59 & 0.4 &0.24 (0.34)&0.4 (0.4)& $0.50\pm 0.52$ &$-0.12$& yes\\ %growth384_B020_Om004_th0.ps AXEL:tmax=.4, Bbarmax=0.6
I2 & 0.28 & 0.006 & 0.61 & 0.4 &0.44 (0.58)&0.4 (0.4)& $0.56\pm 0.29$ &$-0.12$& yes\\% growth384_B020_Om01_th0.ps AXEL:tmax=.5, Bbarmax=0.4
I3 & 0.29 & 0.013 & 0.56 & 1.0 &0.42 (0.64)&0.6 (0.5)& $0.21\pm 0.08$ &$-0.12$& weak\\ %growth384_B020_Om02_th0.ps  AXEL:tmax=.7, Bbarmax=0.5
I4 & 0.30 & 0.019 & 0.59 & 1.0 &0.42 (0.74)&0.7 (1.0)& $0.15\pm 0.07$ &$-0.12$& very weak\\ %growth384_B020_Om03_th0.ps
\hline
% 384k30_B050_Om006_theta
J1 & 0.77 & 0.003 & 0.69 & 0.05&0.20 (0.28) &0.22 (0.22)& $0.10\pm 0.29$ &$-0.15$& very weak\\ %growth384_B050_Om004_th0.ps
J2 & 0.82 & 0.007 & 0.64 & 0.21&0.21 (0.26) &0.32 (0.32)& $0.05\pm 0.10$ &$-0.16$& very weak \\%growth384_B050_Om01_th0.ps
J3 & 0.91 & 0.014 & 0.69 & 0.37&0.23 (0.30) &1.31 (1.31)& $0.02\pm 0.12$ &$-0.15$& no\\%growth384_B050_Om02_th0.ps
J4 & 0.91 & 0.021 & 0.73 & 0.73&0.31 (0.38) &1.36 (1.36)& $0.03\pm 0.10$ &$-0.16$& no\\%growth384_B050_Om03_th0.ps
\hline
\hline
\label{runs3}
\end{tabular}}
\tablefoot{
The colatitude is set to $\theta=0\degr$, corresponding to the pole.
All other quantities are defined in \Tabs{runs1}{runs2}.
}\end{table*}

\begin{figure*}[t!]\begin{center}
\includegraphics[width=0.9\textwidth]{384B002_co30}
\end{center}\caption[]{
Rotational dependency on the active region formation.
We show the vertical magnetic field $B_z/\Beq$ for Runs~A2, B2 and F2
corresponding to $\Co = 0$, $0.0012$, $0.0076$ at the time of the strongest BR.
In the first panel for $\Co = 0$, the value of $\theta$ is insignificant.
}\label{bb3_xy}
\end{figure*}

The energy transfer to larger scales
is reminiscent of an inverse cascade.
\cite{BGJKR14} have speculated that such a cascade might be a
consequence of the conservation of cross helicity, $\overline{\uu\cdot\bb}$,
where the overbar denotes horizontal averaging,
$\bb=\BB-\meanBB$ are the magnetic fluctuations and
$\uu=\UU-\meanUU$ are the velocity fluctuations;
see \cite{RKB11}.
(For horizontal averages, we usually have $\meanUU=0$.)
They studied the production of $\overline{\uu\cdot\bb}$ as a result of a mean
magnetic field along the direction of gravity, so there exists a
pseudoscalar $\grav\cdot\BB_0$ that has the same symmetry properties as
$\overline{\uu\cdot\bb}$ and is also odd in the magnetic field.
A similar result was also obtained by \cite{KKMRSZ03}; see their
Eq.~(11) where they considered inhomogeneous turbulence.
Furthermore, recent work of \cite{ZB18} has shown that the cross
helicity spectrum shows a steep slope at large scales.
This was interpreted as a potential signature of NEMPI-like effects.

\subsection{Influence of rotation}
\label{InfluenceRotation}

The structure of the bipolar regions is strongly influenced by
rotation; see \Fig{bb3_xy}.
Faster rotation causes the magnetic flux concentrations to be
weaker. In some cases, for example in Run~B3
(middle panel of \Fig{bb3_xy}),
the structure of BRs splits into three parts with one
negative and two positive polarities.
In our most rapidly rotating case (right-most panel of \Fig{bb3_xy}), the
structure appears rotated by $90\degr$ with respect to slower rotations.

\Tab{runs2} shows that, as the Coriolis number is increased from 0.0012 to
0.0076, BR formation gets almost entirely suppressed (see Runs~B1 to F4).
At $\theta=0\degr$ (corresponding to the pole), BR formation tends
to be slightly easier; but as the Coriolis number is increased,
the BRs become clearer at intermediate latitudes.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pLambd_co.eps}
\end{center}\caption[]{
Rotational dependency of the growth rate $\lambda$, the peak spectral
magnetic field $(2k_\ast E_{\rm M}^{z,\max})^{1/2}$, and the maximum of
the Fourier-filtered vertical magnetic field $\Bfm$ for colatitudes
$0^\circ$ (black line), $30^\circ$ (blue line), $60^\circ$ (purple
line), and $90^\circ$ (red line).
The error bars are either the errors of the exponential fit
($\lambda$; see \Tab{runs2}), or estimated as $10\%$ of the actual
value.
}\label{pLam_co}\end{figure}

The rotational dependency of the growth rate $\lambda$
for different colatitudes $\theta$ is shown in \Fig{pLam_co}.
For most of the colatitudes, the growth rate first increases for weak
rotation and then gets reduced to around a third of the values for more
rapid rotation.
Even though we cannot visually detect any clear indication of BRs in
the rapidly rotating runs,
the growth rate of magnetic field is still positive.
The magnetic field strength determined from the spectral energy (see
middle panel of \Fig{pLam_co}) shows
actually an enhancement with
rotation. Even for the rapidly rotating cases, the magnetic field
strength in the flux concentrations
is stronger than without any rotation. The strongest values
are achieved with $\Co=0.002$.
For most of the colatitudes, the maxima of the large-scale magnetic
field, as shown in the bottom row of \Fig{pLam_co}, decrease for increasing
rotation, similarly as the $\lambda$.
However, we see the maximum value for $\Co=0.002$ to $0.003$.

The time when the Fourier-filtered magnetic field
and the field of spectral
energy become maximal depends on rotation, but there is no clear
trend visible; see \Tabs{runs2}{runs3}. However there seems to be an indication
that the time becomes longer, as expected for a smaller growth rate.
As the Coriolis number increases, the structures become weaker and it gets more
difficult to discern a clear rotation pattern.

We note here that, unlike in convection, the energy-carrying length scale is
not influenced by rotation, because the driving scale is prescribed through
the forcing function. Similarly, the kinetic energy is also only weakly
influenced by rotation, so $\urms$ decreases only weakly for more rapid
rotation.

\subsection{Dependence on latitude}

As we change the colatitude $\theta$, the growth rates and the
strengths of BRs change.
This is demonstrated in \Fig{bzVStheta}, where
we show $\lambda\tautd$, $(2k_\ast E_{\rm M}^{z,\max})^{1/2}/\Beqz$ and
$\Bfm/\Beqz$
for $\Co=0.0012$ (corresponding to
$2\Omega/\lambda_{\ast0}\approx0.15$)
and different values of $\theta$.
For $\theta=0$, which corresponds to the pole,
the saturation magnetic field strength shows a maximum
at $t/\tautd\approx1$.
For larger values of $\theta$, i.e., closer to the equator,
the maximum is slightly higher (about 0.8) and more long-lived,
e.g., for $0.5\leq t/\tautd\leq2.5$ at $\theta=60\degr$, when
$\meanB_z^{\rm rms}/\Beq$ is above 0.6 and sometimes even 0.8.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pLambd_col}
\end{center}\caption[]{
Summary of the results for the different colatitudes showing the normalized
growth rate of the Fourier-filtered $z$-component of the magnetic field
(top panel) for $B_0/\Beqz = 0.02$,
the averaged magnetic spectrum over $k\leq 4$ (middle panel) and the
maximum of the Fourier-filtered vertical magnetic field $\Bfm$ (bottom panel)
for different values of the Coriolis number and colatitude.
As elsewhere, error bars are either the errors of the exponential fit
($\lambda$; see \Tab{runs3}), or estimated as $10\%$ of the actual
value.
}\label{bzVStheta}\end{figure}

Bipolar structures are still fairly pronounced at $\theta=60\degr$,
i.e., at $30\degr$ latitude; see \Fig{bb3_xy01}.
It is remarkable that for all values of $\theta$,
the inclination of BRs is approximately the same.
The same feature is also seen in the MFS, except that for
$\theta=90\degr$ the structures are always aligned in the $y$ direction.
This may here be the case, too, but the structures are so weak
that this is hard to see.
For somewhat faster rotation, when $\Co>0.01$, corresponding to
$2\Omega/\lambda_{\ast0}\approx1$, the structures disappear.

\begin{figure}[t!]\begin{center}
\includegraphics[width=0.9\columnwidth]{384B002_om004}
\end{center}\caption[]{
Vertical surface magnetic field $B_z/\Beq$ for Runs~C1--C3 and C5 with $\Co = 0.0012$.
}\label{bb3_xy01}
\end{figure}

\subsection{Inclination of BRs}
\label{Inclination}

A systematic orientation of BRs was already seen in the original
papers of \cite{LBKMR2012,LBKR2013a} both for MFS and DNS.
For most of the runs, the BRs are either aligned
with the imposed magnetic field or they are inclined by $45^\circ$.
To compare with the Sun, we map our Cartesian coordinate system
to spherical ones via $(x,y,z)\to(\theta,\phi,r)$, so the
$y$ coordinate points in the toroidal (eastward) direction and
$x$ corresponds to colatitude, which points southward.
This explains the orientations of our surface visualizations
where we plot $-x$ versus $y$; see \Fig{bb3_xy01}.

We usually find that rotation leads to a poleward tilt of the BR.
We return to the question of the inclination angle in
\Sec{InclinedSurfaceStructures} where we study a similar
phenomenon in MFS and show that the inclination angle is then not
an artifact of the domain size.
Note also that the orientation is the same in DNS and MFS.
In fact, the orientation agrees with that found by
\cite{LBKMR2012,LBKR2013a}.
It is interesting to note that the sense of inclination
is the other way around than what is expected based on the buoyant
rise of magnetic flux tubes, which gives rise to Joy's law \citep{CG87}.

The ``anti-Joy's'' law orientation of NEMPI structures is likely a
consequence of the interaction of rotation with the concentration of
flux as opposed to the expansion of flux, which is usually expected as
a flux tube rises through a stratified layer.
A similar phenomenon of a concentration of flux in stratified turbulence
(as opposed to an expansion) has been found in turbulence driven by the
magneto-rotational instability, where this was argued to be the reason
for an unconventional sign of the $\alpha$ effect \citep{BNST95}; see
the more detailed discussion of \cite{BC97}.

\subsection{Dependence on the imposed magnetic field strength}
\label{DependenceFieldStrength}

Rotation weakens the formation of structures for even smaller Coriolis
numbers than those of previous studies \citep{LBKMR2012,LBKR2013a}.
We can therefore increase the efficiency of the formation mechanism
by increasing the imposed magnetic field,
as was suggested by \cite{WLBKR2015}.
The increase in the imposed field
allows the BRs to be formed for even
higher Coriolis numbers, up to the point where
the magnetic field becomes so strong that
the derivative $\dd\Peff/\dd \mean{B}{}^2$ becomes positive and NEMPI cannot be
excited in the domain;
for details, see \cite{BRK2016} and the appendix of \cite{Kemel13a}.

\Tab{runs3} shows simulations of domains located at the pole ($\theta=0\degr$)
and gives their dependence on the magnetic field strength and angular velocity.
As $B_0/\Beqz$ is doubled from 0.07 to 0.14, BR formation becomes slightly
easier: compare model G6 with model H4 (both are for $\Co=0.018$).
Weak BR formation is only possible in model H4.
For even stronger fields, however, this trend disappears.
In model I4 with $B_0/\Beqz=0.3$, BR formation is now very weak and
for $\Co=0.019$, and in model J3 with $B_0/\Beqz=0.9$,
BR formation is impossible---even for $\Co=0.014$; see
\Tab{TCoB0Beq0} for an overview.

\begin{table}[b!]\caption{
Comparison of BR formation as a function of rotation in terms of Co
and imposed magnetic field $B_0/\Beqz$.
}\vspace{12pt}\centerline{\begin{tabular}{l|cccc}
      &     \multicolumn{4}{c}{$B_0/\Beqz$}  \\
\hline
Co    &  0.07 & 0.14  &  0.3  &  0.8  \\
\hline
\hline
0.0   &  yes  & yes   &       & no\\
0.002 &  yes  & yes   &  yes  &very weak\\
0.006 &  yes  & yes   &  yes  &very weak\\
0.012 &  yes  & yes   &  weak &   no   \\
0.018 &   no  & weak &very weak&  no   \\
\hline
\label{TCoB0Beq0}\end{tabular}}
\tablefoot{The values for $\Co$ were taken from \cite{WLBKR2015}.}
\end{table}

The growth rate of the magnetic field
shows a strong decrease for higher rotation rates. This can
be compensated for to some extent by using a stronger imposed magnetic field; see the top
panel of \Fig{pLam_B0}.
For $\Co\approx0.006$, the growth rates with imposed magnetic fields
of $B_0/B_{\rm eq}=0.066$ and $B_0/B_{\rm eq}=0.28$ are indeed higher
than for $B_0/B_{\rm eq}=0.026$.
A similar behavior can be found by looking at the magnetic field
strength determined from the spectral energy (see middle panel of
\Fig{pLam_B0}). For imposed magnetic field strengths between $B_0/B_{\rm
eq}=0.066$ and $0.85$, the magnetic field is not
much influenced by rotation and stays roughly constant above
$\Co\approx0.006$ at a level that is even higher than for smaller
rotation.
The large-scale magnetic field $\Bfm$ for imposed magnetic fields
larger than $B_0/B_{\rm eq}=0.06$ does not show a strong rotational
influence. The large-scale magnetic field is higher for larger imposed
magnetic field and keeps this level for a large range of rotation rates.
This means that a higher imposed magnetic field can indeed prevent
NEMPI from being quenched for larger rotation rates. Even
with a large imposed magnetic field, rotational quenching takes place,
but at larger rotation rates.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pLambd_B0.eps}
\end{center}\caption[]{
Rotational dependency of the growth rate $\lambda$, the peak spectral
magnetic field $(2k_\ast E_{\rm M}^{z,\max})^{1/2}$, and the maximum of
the Fourier-filtered vertical magnetic field $\Bfm$ for various averaged
imposed magnetic field strengths $B_0/B_{\rm eq}=0.026$ (blue line),
$0.066$ (red), $0.14$ (green), $0.28$ (cyan), and $0.85$ (black).
Error bars denote either the errors of the exponential fit
($\lambda$; see \Tab{runs2}), or are estimated as $10\%$ of the actual
value.
}\label{pLam_B0}\end{figure}

\subsection{Effective magnetic pressure}
\label{EffectiveMagneticPressure}

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{b_beq22_Om002_theta0}
\end{center}\caption[]{
Profiles of $\Beq$ and $B_z$ at $z=0$ (i.e., the surface of the turbulent
region), for Run~B1 with $\Co=0.0015$, $\theta=0$ at $t/\tautd=1$
and along $y=0$ (black line).
The red solid line gives the Fourier-filtered profiles of $B_z$.
The dotted orange line gives the values of $\Beq$ through the magnetic
structure and the solid orange lines denotes its Fourier-filtered value.
The blue line represents the value of $\Beq$ based on the volume-averaged
velocity, but the local density.
All values have been normalized by the volume-averaged value of $\Beq$.
}\label{b_beq22_Om002_theta0}\end{figure}

The concentration of magnetic field in a NEMPI scenario is possible
due to a turbulence effect on the effective magnetic
pressure, which is the sum of non-turbulent and turbulent contributions
to the large-scale magnetic pressure.
This effect results in a suppression
of the total (hydrodynamic plus magnetic)
turbulent pressure by the large-scale (mean) magnetic field.
This means that an increase of the mean magnetic field due to the instability
will be accompanied by a decrease of the turbulent pressure
and a reduction of the equipartition field strength, $\Beq$.
This is because the hydrodynamic part of the total turbulent pressure,
$p_{\rm hydro}=\meanrho\urms^2/3=\Beq^2/3$,
as well as the turbulent kinetic energy density,
decrease due to an increase of
the turbulent magnetic energy density as well as
the turbulent magnetic pressure
through tangling magnetic fluctuations \citep{KMR96,RK07}.

\FFig{b_beq22_Om002_theta0} shows the profiles of $\Beq$ and $B_z$
at the surface along $y=0$ at $t/\tautd=1$.
This figure clearly shows that at the location of the maximum of the
vertical magnetic field, the equipartition field $\Beq$ is decreased.
Most of this suppression comes from a local decrease in the turbulent
velocity, while the local density through the structure varies only
little; see the blue line of \Fig{b_beq22_Om002_theta0}.
We attribute this behavior to the operation of NEMPI in the
simulation.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pPeff_co}
\end{center}\caption[]{
Dependence of the minimum value of the normalized effective magnetic
pressure, $\Peff$, on the Coriolis number for different values of the
imposed field strength, $B_0/\Beqz$ (the curves have been labeled
by an average value), for the runs listed in \Tabs{runs2}{runs3}.
Again, error bars are estimated using the maximum difference of the total mean
with the means of each third of the time series.
}\label{peffmin}
\end{figure}

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{ppeff}
\end{center}\caption[]{
Dependence of $\Peff(\beta)$ obtained from \Fig{peffmin} by taking an
average for different values of $\Co$ (filled circles).
The blue line is our standard representation for
$\qpz=32$ and $\betap=0.058$, which corresponds to $\beta_\ast=0.33$,
while the red line is a better fit to the present data giving
$\qpz=13$ and $\betap=0.18$, which corresponds to $\beta_\ast=0.65$.
Note that the data point at $\beta=0.85$ and $\Peff=-0.16$ is well
outside any $\Peff(\beta)$ curve that could fit all the data and
has therefore been discarded as an ``outlier''.
As a comparison, we plot the values of \cite{WLBKR2015} as triangles.
}\label{ppeff}
\end{figure}

Next, we determine the normalized effective magnetic pressure,
$\Peff$, as
\begin{equation}
\Peff=(1-\qp)\,\beta^2/2,
\label{peff}
\end{equation}
where $\qp$ is obtained from \citep{Kemel13a}
\begin{equation}
\qp=-{1\over\mean{\BB}^2} \,
\left(\Delta\overline\Pi_{xx}+\Delta\overline\Pi_{yy}
-\left(\Delta\overline\Pi_{xx}-\Delta\overline\Pi_{yy}\right)\,{\mean{B_x}^2
+\mean{B_y}^2\over\mean{B_x}^2 -\mean{B_y}^2} \right).
\label{q_p}
\end{equation}
Overbars denote here $xy$ averaging and the diagonal components of the
total (Reynolds and Maxwell) stress
tensor, $\Delta\overline\Pi_{ii}$, have been obtained from the DNS as
\EQ
\Delta\overline\Pi_{ii}
=\meanrho\,(\overline{u_i^2}-\overline{u_{0i}^2})
+\half(\overline{\bb^2}-\overline{\bb_0^2})
-(\overline{b_i^2}-\overline{b_{0i}^2}),
\EN
where the subscript $0$ refers to values for $B_0=0$
and lower case letters denote fluctuations, i.e.,
$\uu=\UU-\meanUU$ and $\bb=\BB-\meanBB$.
No summation over the index $i$ is assumed.
In \Fig{peffmin} we show the dependence of the minimum value of
$\Peff$ on the Coriolis number (see \Tab{runs2}) and for
different values of the imposed field (see \Tab{runs3}).
It turns out to be relatively insensitive to the value of $\Co$, but it
drops dramatically with increasing strength of the imposed field.

Given that the values of $\Peff$ shown in \Fig{peffmin} for different
values of $\Co$ are almost the same, we conclude that those coefficients
do not depend on $\Omega$.
We can therefore obtain a new set of parameters that is representative
of our model with coronal layer and for all values of $\Omega$, which
is different from our standard representation without a coronal envelope.
This is shown in \Fig{ppeff}.
The values of the parameters $\qpz$ and $\betap$ fit very well with the
data of \cite{WLBKR2015}, shown as triangles.

The new data can be fitted to \Eq{param_beta} by
determining the position and value of the minimum,
$\betamin$ and $\Pmin$, respectively.
Looking at \Fig{ppeff}, we find $\betamin\approx0.3$ and
$\Pmin\approx-0.12$.
We then obtain the fit parameters as \citep{Kemel12a}
\begin{equation}
\betap=\betamin^2\left/\sqrt{-2\Pmin},\right.\;\;
\betastar=\betap+\sqrt{-2\Pmin}.
\end{equation}
This results in the following new set of parameters:
$\qpz=13$ and $\betap=0.18$, which corresponds to $\beta_\ast=0.65$.
We therefore also present in the next section
MFS results based on our model
with this parameter combination.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pQp_co}
\end{center}\caption[]{
Dependence $q_{\rm p}$, $q_{\rm s}$, and $q_{\rm g}$ on the Coriolis number for different values of the
imposed field strength, $\beta_0=B_0/\Beqz$ (the curves have been labeled
by an average value), for the runs listed in \Tabs{runs2}{runs3}.
Error bars are estimated using the maximum difference of the total mean
with the means of each third of the time series.
}\label{Qp_co}
\end{figure}

Along with $\qp$ and $\Peff$, we also determine $\qs$ and $\qg$ in
\begin{equation}
\Delta\overline\Pi_{ij}=-\qp \delta_{ij} {\mean{\BB}^2\over 2}
+\qs \mean{B_i}\mean{B_j} -\qg {g_i g_j\over g^2} \mean{\BB}^2,
\label{pi-mean}
\end{equation}
resulting in
\begin{equation}
\qs={\Delta\Pi_{xx}-\Delta\Pi_{yy}\over\mean{B_x}^2 -\mean{B_y}^2} ,
\label{q_s}
\end{equation}
\begin{equation}
\qg={1\over\mean{\BB}^2} \,
\left(-\Delta\Pi_{zz}-\qp{\mean{\BB}^2\over 2}
  +\qs\mean{B_z}^2\right),
\label{q_g}
\end{equation}
where ${g_i}$ are components of $\gggg$ which, in our setup, has only a
component in the negative $z$ direction.
In \Fig{Qp_co} we plot the dependencies of $\qp$, $\qs$, and $\qg$
on angular velocity and imposed magnetic field strength.
The values for no and weak rotation are consistent with those
obtained by \cite{WLBKR2015} for a non-roting setup and the same
magnetic field strengths.
Because of the large errors resulting from the strong variation in
space and time, we cannot determine whether the sign is negative
or positive.
For larger imposed magnetic field strengths, the situation becomes more
clear.
There, the errors are significantly larger and $\qp$, $\qs$, and $\qg$
do not strongly depend on rotation.
However, we see a dependence on the imposed magnetic field strength.
In particular, $\qp$ is positive for $B_0/\Beqz\ge0.066$ and
decreases from around $1.5$ to $1.0$ for increasing imposed
magnetic field strength.
Next, $\qs$ is also positive for imposed magnetic fields $B_0/\Beqz\ge0.066$,
but has an inconclusive dependence on the imposed magnetic field.
Finally, $\qg$ is also positive for $B_0/\Beqz\ge0.066$,
it decreases with increasing imposed field until it is nearly zero for
$B_0/\Beqz=0.85$.
For $B_0/\Beqz=0.066$, $\qg$ tends to decrease with increasing angular velocity, but not
for the other magnetic field strengths.
We should keep in mind that the $\qp$ and $\qg$ in \Eq{efforce} are
multiplied by the horizontal averaged magnetic field, which will be
larger for larger imposed magnetic fields.
We also look at the latitudinal dependencies of $\qp$, $\qs$, and $\qg$;
see \Fig{Qp_theta}. We do not find any latitudinal dependence of
these coefficients, mostly because the errors are so large.
In conclusion, we cannot explain the rotational dependence
found in the DNS with just the rotational dependence of $\qp$, $\qs$,
and $\qg$.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pQp_theta}
\end{center}\caption[]{
Dependence $q_{\rm p}$, $q_{\rm s}$ and $q_{\rm g}$ on the colatitude
$\theta$ for Runs~C1--C3 and C5 with Co=0.0023 and $B_0/\Beqz=0.026$.
Error bars are estimated using the maximum difference of the total mean
with the means of each third of the time series.
}\label{Qp_theta}
\end{figure}

\section{MFS results}

\subsection{General aspects}

We now present mean-field calculations with a coronal envelope.
In all cases, we use $\nuT=\etaT=10^{-3}\cs H_\rho$ and $\urms=0.1\cs$,
corresponding to $\kf H_\rho=33$, which is similar to the value
of $30$ used in the DNS.
We first consider the same domain size as in the DNS, i.e., $(2\pi)^2\times3\pi$.
To alleviate finite domain size effects, we also consider a wider and
deeper domain, but with a smaller coronal part that is reduced by half.
We thus also consider $L_x=L_y=6\pi$, $L_z=4\pi$, and $z_{\rm top}=\pi$.

The MFS lack small-scale turbulent motions, so fewer mesh points
can be used.
However, to resolve the vertical density contrast of around 12,000, we
used 288 mesh points in the $z$ direction.
For our domains, are used $192^2\times288$ meshpoints, but we found no
differences in the results when using $96^2\times288$ meshpoints.
For the larger domains, we used $384^3$ meshpoints, but again, with
fewer meshpoints the results would have been sufficiently accurate.
In all cases, we use $B_0/\Beqz=0.1$, because the DNS discussed in
\Sec{DependenceFieldStrength}
showed that for approximately this value, the range in rotation rates over
which NEMPI is still excited would be maximized.

Although we have found that the parameters $\qpz\approx13$ and
$\betap\approx0.65$ describe the DNS best, there are reasons to consider
also other choices.
First, there is no good reason why the parameters $\qpz$ and $\betap$
are so different in different circumstances.
One would have expected them to reflect properties of the turbulence
which is similar in all the different cases.
Second, the response for vertical and horizontal magnetic fields turns
out to be different; see Fig.~7 of \cite{LBKR2014}.
There are also other differences between MFS and DNS that we address
below.
Thus, we cannot expect the two approaches to agree. One
objective is therefore to find out just how well the MFS perform relative
to the DNS.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{presults_Omdep}
\end{center}\caption[]{
Dependence of $\lambda/\lambda_{\ast0}$ on $2\Omega/\lambda_{\ast0}$
for $B_0/\Beqz=0.1$ and $\theta=0$ for the MFS using the large (black
line) and small domains (blue solid line), as well as the DNS for
neighboring magnetic field strength of $B_0/\Beqz=0.066$ and 0.14
(dashed and solid red lines, respectively).
The fat blue line denotes the case of a small domain
using $a_{\rm g}=-0.4$ instead of $0$.
The green dashed line denotes the DNS of \cite{LBKR2013a} without coronal
layer and $B_0/\Beqz=0.05$.
The orange line refers to MFS in a small domain with $\qpz=13$ and
$\betap=0.18$, so $\beta_\ast=0.65$.
The blue dashed line adjacent to the blue solid line denotes the results
obtained when neglecting the derivative term of the profile function
$\Theta_w(z)$.
}\label{presults_Omdep}
\end{figure}

\subsection{Growth rates}

In the DNS, the value of $\lambda/\lambda_{\ast0}$ is found to drop by about a
factor of five as $2\Omega/\lambda_{\ast0}$ increases from 0.2 to 3;
see \Fig{presults_Omdep}.
We compare the MFS for $B_0/\Beqz=0.1$ with the DNS at $B_0/\Beqz=0.066$
and $0.14$; see \Tab{runs2}.
We also compare with the DNS of \cite{LBKR2013a} without coronal envelope
using $\beta_\ast=0.33$ and, as in \cite{LBKR2013a}, the value
$\beta_\ast=0.75$, which was suitable for one of their sets of MFS.
This now explains the 2.3 times larger ratio
of $\lambda/\lambda_{\ast0}$ compared to their Fig.~2.
In addition, however, the growth rates of their DNS were incorrectly
scaled by a factor $(\urms/\cs)(\kf/k_1)\approx0.1\times30=3$, which is
now corrected; see the green line of \Fig{presults_Omdep}.
Thus, even in the absence of a corona, the growth rates were
by a factor of about seven larger in the MFS than in the DNS; see
\App{Comparison_Los_etal13} with a corrected version of Fig.~2 of
\cite{LBKR2013a}.

Again, the growth rates of the large-scale instability
in the MFS are significantly larger than those in the DNS.
We speculate that this could be caused by a partial cancelation
(i.e., a decrease of the effective magnetic pressure gradient)
from the $q_{\rm g}$ term in \Eq{efforce}.
To illustrate this possibility, we have overplotted in \Fig{presults_Omdep}
a case with $a_{\rm g}=-0.4$; see \Eq{adef}.
Given that in \Eq{efforce}, only the $\qp$ term comes with a 1/2 factor,
the effective magnetic pressure gradient in the $z$ direction is reduced.
Thus, $\qp/2$ is replaced by $\qp/2+\qg=(1/2+a_{\rm g})\,\qp$, so $\qp$ is
scaled by a factor $1+2a_{\rm g}=0.2$ with respect to the $z$ direction.
We see that the growth rate is now suppressed, but also the maximum
rotation rate for which NEMPI can operate is reduced.
Obviously, a more accurate modeling requires a more detailed knowledge
of the actual form of $q_{\rm g}$, which is likely to be different
from that of $q_{\rm p}$.
We should also point out that, when
calculating $\partial_z q_{\rm g}(\beta)$ in \Eq{efforce},
one of the resulting terms involves the gradient of $\Theta_w(z)$
in \Eq{parameterization}. We have verified that neglecting this term
causes only a very minor change in
the resulting growth rates; cf.\ the solid and dashed blue lines in
\Fig{presults_Omdep}.

Let us discuss the non-monotonic behavior of the
growth rate of the magnetic field as a function of the Coriolis
number; see \Fig{presults_Omdep}.
In corresponding DNS without coronal layer \citep{Jabbari2014},
the increase in the growth rate
at faster rotation rates ($\Co>0.1$ or $2\Omega/\lambda_{\ast0}>10$)
has been explained as a result of large-scale dynamo action;
see Fig.~8 of \cite{Jabbari2014}.
In particular, at larger rotation rates, kinetic helicity is produced by a
combined effect of uniform rotation and density stratified turbulence.
It results in the excitation
of an $\alpha \Omega$ or $\alpha^2\Omega$ dynamo instabilities
and the generation of a large-scale magnetic field.
This causes an increase of the growth rate at larger Coriolis numbers,
which is also observed in the DNS of
\cite{LBKR2013a} and \cite{Jabbari2014}.
This implies that two different instabilities are excited
in the system, i.e., NEMPI at low Coriolis numbers
and the mean-field dynamo instability at larger values of the
Coriolis numbers. This causes a non-monotonic behavior of the
growth rate of magnetic field as the function of the Coriolis
number, observed in \Fig{presults_Omdep}.

A non-trivial evolution of the magnetic field
in rotating turbulence with a coronal envelope
is caused for the following reasons.
In an earlier study by \cite{LBKMR2012}
with rotation, no coronal envelope, and
an imposed horizontal magnetic field,
an analytical expression for the growth rate of NEMPI
has been derived in the framework
of a mean-field approach.
During the magnetic field evolution in the presence of a coronal
envelope, as in the simulation of \cite{WLBKR2013,WLBKR2015} and the
present ones, there is a change in the direction of
the large-scale magnetic field from horizontal at $t=0$ to
nearly vertical after about one turbulent diffusive time.
Therefore, in turbulence with a coronal envelope
one can expect to find a mixture of
effects caused by the horizontal and vertical magnetic fields.

The growth rates and saturation mechanisms of NEMPI for horizontal and
vertical fields are very different \citep[see review by][]{BRK2016}.
For horizontal fields, NEMPI saturates rapidly
in the nonlinear stage of the magnetic field evolution
due to the ``potato-sack" effect.
This means that a local increase of the magnetic field
causes a decrease of the negative effective magnetic pressure,
which is compensated for by enhanced gas pressure.
This leads to an enhanced gas density, so the gas
is heavier than its surroundings and sinks.
This effect removes horizontal magnetic flux structures from regions
in which NEMPI is excited.
For a vertical magnetic field, the heavier fluid moves
downward along the field without affecting the flux tube, so
that NEMPI is not stabilized by the potato sack effect.
In this case the operation of NEMPI results
in the formation of strong concentrations \citep[see][]{BKR2013,BRK2016}.

In the non-rotating cases,
the growth rates of the large-scale magnetic field measured in DNS
with coronal envelope and horizontal initial field \citep{WLBKR2015},
and with vertical magnetic field but without corona \citep{BKR2013}, are the same.
This is an indication that the growth rates of the large-scale magnetic field
in the simulations with coronal envelope are determined by the evolution
of the vertical field.
On the other hand,
the fact that the BRs dissolve after a few turbulent diffusion
times is more similar to the behavior of NEMPI with a horizontal imposed
magnetic field.
Therefore, the evolution of the magnetic structures
in the system with
coronal envelope is non-trivial and cannot easily be described with our
current mean-field models.
Therefore, also explaining the rotational dependency of growth rates
in this setup suffers from the mixture of effects.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{plam_theta}
\end{center}\caption[]{
Dependence of $\lambda/\lambda_{\ast0}$ on $\theta$ for
$B_0/\Beqz=0.1$ and $\Co=0.006$.
}\label{plam_theta}
\end{figure}

\subsection{Latitudinal dependence}

At slower rotation, a decrease in $\lambda/\lambda_{\ast0}$ can be seen
as $\theta$ increases from $\theta=0$ at the poles to $\theta=90\degr$
at the equator; see \Fig{plam_theta}.
In both plots we also show the growth rates for the larger
domain, which are found to be enhanced by a factor of two when
$2\Omega/\lambda_{\ast0}\approx1$.
These values are about an order of magnitude larger than those
for the DNS, but have otherwise a similar functional dependence
on $\Omega$ and $\theta$.
The reason for the difference between DNS and MFS is not entirely clear.
It is possible that the mean-field parameter $\beta_\ast$ is smaller
than what was previously found for simulations with coronal envelope.
There is also the possibility that $\beta_\ast$ decreases with increasing
angular velocity, which is what R\"udiger (private communication) found,
although our present simulations presented in \Fig{peffmin} and earlier
ones of \cite{LBKR2013a} did not give such indications.

\subsection{Inclined surface structures}
\label{InclinedSurfaceStructures}

Next, we show slices of $\meanB_z(x,y,0,t_\ast)$ along the surface $z=0$
at a chosen time $t_\ast$ during the linear growth phase of NEMPI for three
values of $\Co$ using domain sizes of $(2\pi)^2\times3\pi$
(\Fig{pslice_comp2}) and $(6\pi)^2\times4\pi$ (\Fig{pslice_comp}).
In the linear phase, when the magnetic field fluctuations are still
growing exponentially in time, only relative values are of physical
interest.
We therefore present in the following images where the magnetic field
is normalized by the maximum value.
As in the DNS, the imposed magnetic field points in the $y$ direction.
It turns out that rotation not only tends to make the structures inclined
relative to the direction of the imposed magnetic field,
but it also leads to higher wavenumbers of the structures.
\FFig{pslice_comp} shows that the number of nodes in the $x$ direction,
which is perpendicular to the magnetic field, remains about constant
($k_x L_x/2\pi=4$), while that in the $y$ direction along the magnetic
field increases from $k_y L_y/2\pi=1$ (for $\Co=0$) to 2 (for $\Co=0.006$)
and 4 (for $\Co=0.018$).

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pslice_comp2}
\end{center}\caption[]{
Slices of $\meanB_z(x,y,0,t_\ast)$ through the surface $z=0$
at times $t_\ast$ during the linear growth phase of NEMPI
for $B_0/\Beqz=0.1$ and three values of $\Co$ using the smaller
domain size of $(2\pi)^2\times3\pi$.
In each panel, the magnetic field is scaled to the maximum value.
}\label{pslice_comp2}
\end{figure}

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pslice_comp}
\end{center}\caption[]{
Similar to \Fig{pslice_comp2}, but for the large domain,
$(6\pi)^2\times4\pi$.
}\label{pslice_comp}
\end{figure}

In both the DNS and the MFS, the orientation of the inclination is the same
and it is opposite to what is seen in Joy's law.
The runs presented here apply of course only to the poles, but even
at lower latitudes (e.g., at $30\degr$ latitude, corresponding to
$\theta=60\degr$) do we find the same anti-Joy's law orientation of
the tilt.
This is shown in \Fig{pslice_comp80}, where we compare two runs with
$\theta=80\degr$ (close to the equator) for a section of the large
domain and the full section of the smaller domain, as well as a run
with $\theta=90\degr$ (at the equator).
At the equator, the inclination angle with respect to the toroidal
direction is $90\degr$, which agrees with what was found in
\cite{LBKMR2012}.
However, slightly away from the equator, at $\theta=80\degr$, the
inclination angle is already $45\degr$.
This is not an artifact of having chosen a small domain, because even
for a three times larger domain, the same inclination angle is found.
We thus remain puzzled about this finding and hope to be able to
return to it as further numerical and analytic results can be obtained
and assessed.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pslice_comp80}
\end{center}\caption[]{
Similar to \Fig{pslice_comp2}, but for two runs with
$\theta=80\degr$ for a 1/3 section of the large domain
and the full section of the smaller domain,
as well as a run with $\theta=90\degr$.
}\label{pslice_comp80}
\end{figure}

One might speculate that the reason for the difference to Joy's law has
to do with the expansion of rising structures whereas NEMPI structures
are caused by contraction, which leads to the opposite tilt.
Thinking again of possible applications to the Sun, one may therefore
wonder whether the effect of flux concentrations in NEMPI, which must also
be responsible for causing the tilt, operate on relatively small scales
and might be responsible for causing sunspot rotation
\citep{Eve09,Kem10,Pev12,Stu15}, for example.
In a distributed solar dynamo scenario, the tilt of active regions is
primarily caused either by differential rotation \citep{Bra05} or simply
by the sign of the mean latitudinal field $\meanB_\theta$ relative to
that of the azimuthal field $\meanB_\phi$ \citep{Jabbari15}.

\subsection{Emergence of solitary structures}

The magnetic field patterns in \Figss{pslice_comp2}{pslice_comp80}
are more or less regular.
This is basically because those pictures were taken from the linear
phase of the run.
In \Fig{pxy_slices_1h} we show a visualization of $\meanB_z$ along with
horizontal flow vectors $(\meanU_x,\meanU_y)$ through the surface for
an arbitrarily chosen time $t_\ast/\tautd=40$ during the saturated state
for the large domain using $\Co=0.006$.
We can now clearly see solitary structures in the form of isolated spots.
However, with a few exceptions, most of these structures lack the distinct
bipolarity seen in the DNS.
Note also that the mean flow is mostly circular around each spot rather
than a convergent inflow, as seen in the DNS.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pxy_slices_1h}
\end{center}\caption[]{
A slice of $\meanB_z(x,y,z_\ast,t)$ (color coded) together with
velocity field vectors (white streak lines) through $z_\ast=0$ at
$t_\ast/\tautd=40$ during the saturated state for the large domain
using $\Co=0.006$.
The white horizontal line through $x=0$ marks the position
of a BR near $y=0$, which is discussed separately.
The ellipse marks the position of the BR discussed in the text.
}\label{pxy_slices_1h}
\end{figure}

In \Fig{pxy_slices_1h} we mark with a white horizontal line in
the toroidal direction through $x=0$ the position of a BR near $y=0$.
Unlike the DNS, the separation of the two polarities is rather large
(about $3\,H_\rho$) and there are other spots in almost the same
distance.
Thus, it is not clear that these two polarities are connected to each
other.
There is also no clear indication of BRs.
To examine this further, we present in the next section a side view of
this BR.

\subsection{Side view of BRs}

In \Fig{pyz_slices_1h} we show a longitudinal cross-section of
the magnetic field through $x_\ast=0$ at the same time as
\Fig{pxy_slices_1h} during the saturated state.
Magnetic flux concentrations are seen to occur at a depth of
$z/H_\rho\approx-6$, which is well below the surface.
Near the surface, on the other hand, there is only a relatively
small number of vertical magnetic flux structures
that seem to close upon themselves over relatively large horizontal distances.
We recognize the positive and negative polarities at
$y/H_\rho\approx\pm1.5$ and the negative one at $y/H_\rho\approx-8$
in both \Figs{pxy_slices_1h}{pyz_slices_1h}.
Conversely, over short distances, bipolar magnetic flux
structures separate above the surface, which is consistent
with them being the result of a localized subduction of
a horizontal flux structure.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pyz_slices_1h}
\end{center}\caption[]{
A slice of $\meanB_z(x_\ast,y,z,t)$ (color coded) together with
magnetic field vectors (white streak lines) through $x_\ast=0$
at an arbitrarily chosen time during the saturated state for
the large domain.
The surface at $z=0$ is shown as a white horizontal line.
}\label{pyz_slices_1h}
\end{figure}

To compare with DNS results, we show in \Fig{pyz_slice} a similar plot
of $B_z$ together with Fourier-filtered magnetic field vectors in the
same plane.
A major difference to \Fig{pyz_slices_1h} is the absence of
significant horizontal field in the deeper parts.
However, since this horizontal field in the MFS is so deep down
($z/H_\rho\la-8$), it is unclear whether it plays any role in
explaining the difference in, for example, the growth rates
between DNS and MFS seen in \Fig{presults_Omdep}.
On the other hand, in the deeper parts of the MFS, there are magnetic
structures of significant strength, which are not so prominent in the DNS.
This is an important difference that would affect global comparisons of,
for example, the growth rates of structures shown in \Fig{presults_Omdep}.

\begin{figure}[t!]\begin{center}
\includegraphics[width=\columnwidth]{pyz_slice}
\end{center}\caption[]{
Comparison with DNS showing $B_z(x_\ast,y,z,t)$ (color coded) together
with vectors of Fourier-filtered magnetic field vectors ($k<\kf/2$)
superimposed for Run~A4 at the time $t/\tautd=0.91$ through $x_\ast=0.5$.
}\label{pyz_slice}\end{figure}

\section{Discussion}

Our calculations have been performed using idealizing circumstances
such as forced turbulence and an isothermal equation of state.
This is in many ways different from turbulence in the Sun,
which is driven by convection.
Nevertheless, some tentative conclusions can be drawn regarding
possible applications to sunspot formation.
In the surface layers of the Sun, the turnover time $\tauto=\HP/\urms$
based on the pressure scale height $\HP$ is around $5\min$.
Assuming $k_1\HP=1$, the turbulent diffusive time scale is
related to this via
\EQ
\tautd=\HP^2/\etatz=3\kf\HP^2/\urms=3\kf\HP\tauto.
\EN
In the simulations, we have $\kf\HP\approx\kf/k_1=30$,
so $\tautd$ would be about 90 times longer than $\tauto$,
i.e., about 8 hours.
This would appear suitable in view of applications to the Sun as well,
where the formation time of sunspots is of a similar order.

Although the presence of our simplified corona has the effect of allowing
BRs to form that reach a significant fraction of the equipartition value
with respect to the turbulence, these structures can no longer form when
the Coriolis number exceeds a critical value of about 0.02.
This value is rather small.
However, given that the growth rate of NEMPI does not scale
with the inverse turnover time $\tauto^{-1}$, but with the
turbulent-diffusive
time $\tautd^{-1}$, a meaningful measure of the rotation rate in this
context could also be the square root of the {\it turbulent} Taylor number,
$\Tat^{1/2}=2\Omega/\nut k_1^2=3\,(\kf/k_1)^2\Co$,
where we took into account that the turbulent viscosity $\nut$ is equal
to the turbulent magnetic diffusivity $\etat$ \citep{YBR03,KR94}
and given by $\etat\approx\urms/3\kf$ \citep{SBS08}.
The values of $\Ta^{1/2}$ are typically of the order of 10 when NEMPI
begins to be suppressed.
Using our estimate of $\tautd=8\,{\rm hours}$ for the solar surface,
we find $\Ta^{1/2}=0.2$, which is well below our critical value of 10.

In the standard mixing length theory of convection, the value of $\kf\HP$
is estimated to be around 6--7 \citep{Kemel13a}, but it is about $30$ in
the present simulations.
Earlier work showed that NEMPI would not work for $\kf\HP$ much below 15
\citep{BKKR12}.
On the other hand, the actual value in the Sun is unclear given that
the findings of \cite{HDS12} did not confirm turbulent velocities
in the Sun at the expected levels.
A possible resolution to this problem might be the idea that the relevant
scales of the energy-carrying eddies is much smaller than the inverse
pressure scale height \citep{Bra15}.
This would also help making NEMPI more powerful.
However, this issue is controversial in view of results by \cite{Greer15},
which showed that the turbulent flows in the Sun might actually be
just as large as assumed in standard mixing length theory.
Thus, the possibility of NEMPI being responsible for the production
of sunspots is being favored particularly in the scenario envisaged
by \cite{HDS12}.

An additional complication is that at the solar surface, radiation
plays an important role.
Mean-field models with radiation transport \citep{PB18} have shown
that the relevant length scale of NEMPI can drop significantly
below the value found for isothermal and isentropic stratifications.
It is possible, however, that this result is a consequence of not
having included the convective flux in such a model.
In the Sun, virtually 100\% of the energy is transported by convection
almost immediately beneath the surface, so radiation should be
completely unimportant below the surface.
In addition, ionization dynamics can strongly exaggerate the effects
of cooling near the surface.

\section{Conclusions}

Our work has confirmed that NEMPI cannot be exited at Coriolis numbers
above a critical value that can be as low as $0.02$ or so.
The presence of an upper coronal layer was previously found to make the
appearance of structures more prominent.
However, rotation seems to affect the growth
rates more strongly with than without a coronal envelope.
In the bulk of the solar convection zone, the Coriolis number is of the
order of unity and above, but this is not the case in the surface layers,
where the convective time scale is much shorter than the solar rotation
period of 25 days.
So this may not really be a problem for applications of NEMPI to sunspot
formation.

A more severe problem for astrophysical applications of NEMPI are the
moderate magnetic field strengths that can presently be achieved with
NEMPI.
This suggests that some essential physics is still missing.
An important ingredient of sunspots physics is convection and its
suppression in the presence of magnetic fields.
A number of aspects such as radiation and ionization physics, taken in
isolation, have not yet produced more favorable conditions for NEMPI
\citep{BhatBrandenburg15,PB18}.

It is important to realize that the difficulty in explaining the
spontaneous formation of sunspot-like magnetic flux concentrations is
not really alleviated by invoking the rising flux tube scenario.
The problem here is that any flux tube rising from some depth to the
surface will expand and therefore weaken and so some mechanism for
reamplification is needed.
Invoking therefore a suitable model for convection,
or possibly the inclusion of a magnetic
suppression of turbulent radiative diffusion as
suggested by \cite{KitchatinovMazur2000},
might be an important next step.

\begin{acknowledgements}
J.W.\ acknowledges funding by the Max-Planck/Princeton Center for
Plasma Physics and funding from the People Program (Marie Curie
Actions) of the European Union's Seventh Framework Programmed
(FP7/2007-2013) under REA grant agreement No.\ 623609.
This work has been supported in part by
the NSF Astronomy and Astrophysics Grants Program (grant 1615100),
the Research Council of Norway under the FRINATEK (grant 231444),
the Swedish Research Council (grant 2012-5797),
and the University of Colorado through its support of the
George Ellery Hale visiting faculty appointment.
I.R.\ acknowledges the hospitality of NORDITA
and Max Planck Institute for Solar System Research in G\"ottingen.
We acknowledge the allocation of computing resources
provided by the Swedish National Allocations Committee at the Center for
Parallel Computers at the Royal Institute of Technology in Stockholm.
This work utilized the Janus supercomputer, which is supported by the
National Science Foundation (award number CNS-0821794), the University
of Colorado Boulder, the University of Colorado Denver, and the National
Center for Atmospheric Research. The Janus supercomputer is operated by
the University of Colorado Boulder.
Additional simulations have been carried out on supercomputers at
GWDG, on the Max Planck supercomputer at RZG in Garching, in the
facilities hosted by the CSC---IT Center for Science in Espoo,
Finland, which are financed by the Finnish ministry of education.
\end{acknowledgements}

\bibliography{ref}
\bibliographystyle{aa}

\appendix

\section{Comparison of growth rates in DNS and MFS of \cite{LBKR2013a}}
\label{Comparison_Los_etal13}

In \cite{LBKR2013a}, the growth rates were accidently scaled by
a factor $(\urms/\cs)(\kf/k_1)\approx0.1\times30=3$.
In addition, they used $\beta_\ast=0.75$, which was suitable for one of
their sets of MFS, but not for the other.
Therefore, the growth rates of their MFS exceeded those of their DNS by
a factor of about seven.
The corrected version of their Fig.~2 is shown in
\Fig{presults_Omdep_Los_etal13}.

\begin{figure}[h!]\begin{center}
\includegraphics[width=\columnwidth]{presults_Omdep_Los_etal13}
\end{center}\caption[]{
Corrected version of Fig.~2 of \cite{LBKR2013a} showing the
dependence of $\lambda/\lambda_{\ast0}$ on $2\Omega/\lambda_{\ast0}$
for DNS (red dashed line), compared with MFS~(i) where $\qpz=20$ and
$\betap=0.167$ (black solid line), and MFS~(ii) where $\qpz=32$
and $\betap=0.058$ (blue dash-dotted line).
}\label{presults_Omdep_Los_etal13}
\end{figure}

\end{document}

